DOCUMENTOS DE TRABAJO

Is the Information Channel of Monetary Policy Alive in Emerging Markets?

Mariana García-Schmidt



BANCO CENTRAL DE CHILE







La serie Documentos de Trabajo es una publicación del Banco Central de Chile que divulga los trabajos de investigación económica realizados por profesionales de esta institución o encargados por ella a terceros. El objetivo de la serie es aportar al debate temas relevantes y presentar nuevos enfoques en el análisis de los mismos. La difusión de los Documentos de Trabajo sólo intenta facilitar el intercambio de ideas y dar a conocer investigaciones, con carácter preliminar, para su discusión y comentarios.

La publicación de los Documentos de Trabajo no está sujeta a la aprobación previa de los miembros del Consejo del Banco Central de Chile. Tanto el contenido de los Documentos de Trabajo como también los análisis y conclusiones que de ellos se deriven, son de exclusiva responsabilidad de su o sus autores y no reflejan necesariamente la opinión del Banco Central de Chile o de sus Consejeros.

The Working Papers series of the Central Bank of Chile disseminates economic research conducted by Central Bank staff or third parties under the sponsorship of the Bank. The purpose of the series is to contribute to the discussion of relevant issues and develop new analytical or empirical approaches in their analyses. The only aim of the Working Papers is to disseminate preliminary research for its discussion and comments.

Publication of Working Papers is not subject to previous approval by the members of the Board of the Central Bank. The views and conclusions presented in the papers are exclusively those of the author(s) and do not necessarily reflect the position of the Central Bank of Chile or of the Board members.

Documentos de Trabajo del Banco Central de Chile Working Papers of the Central Bank of Chile Agustinas 1180, Santiago, Chile Teléfono: (56-2) 3882475; Fax: (56-2) 38822311 Documento de Trabajo Nº 1017 Working Paper N° 1017

Is the Information Channel of Monetary Policy Alive in Emerging Markets?*

Mariana García-Schmidt[†] Central Bank of Chile

Abstract

Central Bank policy decisions affect the economy not only by influencing market conditions through its market interventions but also by shaping the people's expectations of economic conditions via the announcement of those decisions. This paper studies how forecasts of inflation and output growth respond to unexpected policy rate decisions using datasets for Brazil and Chile that satisfy three conditions: high enough frequency, short-term horizons, and the same source for the dependent and independent variables. The results show that inflation and output forecasts increase in the short run after an unexpected increase in the policy rate, which supports the existence of an information shock behind the monetary policy decision. These results can be explained by a baseline Neo-Keynesian model only when the interest rate provides information about shocks other than the monetary policy shock.

Resumen

Las decisiones de política monetaria afectan a la economía no sólo al influir condiciones de mercados al cambiar tasas, sino también al cambiar las expectativas futuras afectadas por los anuncios de estas decisiones. Este artículo estudia cómo las expectativas de inflación y crecimiento responden ante cambios inesperados de la tasa de política monetaria utilizando datos para Brasil y Chile que satisfacen tres condiciones: alta frecuencia de recolección, horizontes de corto plazo para las expectativas y fuente común para la medición de variables dependientes e independiente. Los resultados muestran que las expectativas de inflación y producto aumentan en el corto plazo luego de un aumento inesperado en la tasa de política, lo que respalda la existencia de un shock de información detrás de esta decisión de política. Estos resultados pueden explicarse mediante un modelo básico neokeynesiano sólo cuando la tasa de interés proporciona información sobre shocks distintos del shock de política monetaria.

^{*}I thank Álvaro Castillo, Giancarlo Acevedo, Daniela Mondaca and Almendra Almonacid for their excellent research assistance. For very useful comments and suggestions I thank Martin Uribe, Michael Woodford, Stephanie Schmitt-Grohé, Argia Sbordone, Carlos Carvalho, Domenico Giannone, Jón Steins-son, Pablo Otonello and seminar participants at Columbia University, the Federal Reserve Bank of New York, the Central Bank of Chile, among others. The views expressed are those of the author and do not necessarily represent those of the Central Bank of Chile or its board members.

1 Introduction

Central Bank policy decisions affect the economy not only by influencing market conditions through its market interventions but also by shaping the people's expectations of economic conditions via the announcement of those decisions. There is important debate in the literature about the kind of shocks that affect the economy because of monetary policy decisions, mainly because their effects on economic variables, including expectations, are very different. On the one hand, there is the old-fashioned monetary shock, which is the one that —all-else-being-equal— increases interest rates. This could be separated into a shock to current or future interest rates. Still, in both cases, a positive shock should diminish the incentive to consume and decrease the demand (and ultimately output) and inflation and all their expectations.¹

On the other hand, there is a relatively new information shock, which shows reverse responses. This is so because, in general, a higher-than-expected monetary policy rate (MPR) would signal higher-than-expected demand and hence would increase expected inflation and output. Such effects are opposite to the policy's intended effects. For example, a policy meant to stimulate aggregate demand by lowering the interest rate may reveal that the central bank expects demand to be low, reducing people's forecast of aggregate demand and their willingness to spend rather than increasing it.

This paper contributes to the debate in the literature about the occurrence of monetary and information shocks when making the MPR decision. The empirical part identifies one shock during the MPR announcement and argues that if the shock affects expectations of realized but unknown variables or variables in the very short term, there is evidence of an information shock. If there were no information shocks, there should be no reaction of expectations in the past and near term. In contrast, in the longer term, expectations could react because of information or the effect of the monetary shock. If there were only monetary shocks, that reaction should be negative because a higher-than-expected interest rate decreases incentives to consume, inflation, and output. If there are also information shocks, that reaction depends on the timing of the information provided by the MPR decision. If it is about the short term, long-term expectations would not react, but if the information is about the future, there could also be a positive reaction. Ultimately, the

¹Another literature has proposed that increases in the interest rate should drive up inflation even when these changes do not convey information that the central bank knows about the economy. This view has been called "Neo-Fisherian" and can be found on the internet, for example, in Smith (2014), Williamson (2014), and Cochrane (2015). García-Schmidt and Woodford (2019) show that the theory does not support this view.

movement of longer-term expectations could not be informative about the information channel.

This paper finds that short-term inflation expectations and past and short-term output expectations react, on average, positively to higher-than-expected changes in the MPR. These results are based on the reaction of inflation expectations of the main price indexes and of GDP growth expectations of Brazil and Chile, and on the reaction of the expectations of Industrial Production (IP) growth in Brazil. In the medium-to-long term, expectations show mixed results, but in all cases, the reaction is smaller than in the short term. These findings are consistent with an active information shock of monetary policy.

Then, this paper develops the simplest possible Neo-Keynesian model to disentangle which shock can be behind the short-term positive responses found in the empirical section. The model includes three types of current shocks, namely a monetary shock, a natural rate of interest shock, a cost-push shock, and three types of news shocks that signal future realizations of the same shocks. The results show that the surprise in the MPR has to give enough information about the natural rate (current or future) to get positive revisions of current output and inflation to an unexpected positive realization of the MPR.

The monetary and information shocks are hard to identify because they happen simultaneously and affect the same variables. The literature has implemented several methods to separate these, which include co-movement between interest rates and stocks (Jarociński and Karadi, 2020; Cieslak and Schrimpf, 2019), orthogonality to the central bank's information set (Miranda-Agrippino and Ricco, 2021), and differentiated effects in different interest rates (Bu et al., 2021) among others. There is also active discussion about whether information shocks still exist or they were a thing of the past (Hoesch et al., 2023 and D'Agostino and Whelan, 2008 updating the analysis of Romer and Romer, 2000, who found previous asymmetries of information between the Federal Reserve and a survey) or are endogenous reactions to other information and so the effects disappears when controlling for different variables (Bauer and Swanson, 2023).

This paper makes several contributions to the debate about the responses of expectations to monetary policy surprises and the existence of the information channel. First, and in contrast to the vast majority of the literature, the dataset presents several advantages of studying the reaction of expectations to monetary versus information shocks. The most important characteristics that satisfy both datasets are three: (i) In comparison to other surveys (such as Blue Chip in the U.S.), these datasets have closer pre- and postmeeting measurements, which is very important to avoid getting spurious effects (Bauer and Swanson, 2023)). (ii) It includes expectations of past and short-term variables, which is important because these are the ones that do not move as a response to a monetary shock and hence identify an information shock. (iii) The datasets include measures of the shock and the expectation reaction from the same source. How the expectations of an agent react to a shock depends on what the agent expected, not what others expected.

One can argue that if the measures are based on market data, there should be market consensus, but this is hard to argue when the measures of the dependent and independent variables come from different sources.² Additional characteristics include the fact that The shock and revision measurements are direct measures of expectations, instead of relying on approximations of one or both variables using different methods (Acosta, 2023; Faust et al., 2004 among others). Finally, and in the case of Chile, this paper is, to the best of my knowledge, the only one in the literature that studies using individual responses of a group of institutions, which is very beneficial in this particular case since the monetary authorities do not surprise the market very often.

Third, we contribute to analyzing the information channels in two emerging markets, which are very scarce in the current literature. The reason for using Chile and Brazil was their datasets' quality. Still, using these two countries is very beneficial since both are inflation targeters and have similar monetary policy frameworks.

Fourth, it highlights the importance of the information channel not only on the whole communication delivered on the day of the monetary policy meeting but also on the information about only the decision of the MPR. This is important because it implies that care must be taken not only about the information released during policy meetings but also about the policy decision itself. The rest of the literature uses, in general, a surprise component that includes the change in the rate and information given about the short and medium run (Faust et al., 2004; Jarociński and Karadi, 2020 among many others).

This paper relates directly to the empirical literature that studies the effect of surprises in monetary policy on revisions of expectations and discusses the existence of information shocks. Before 2020, the literature was scarce and included only a handful of papers. It started with Romer and Romer (2000), who found a reaction of forecasts because of

 $^{^{2}}$ Kuttner (2001) and Gürkaynak et al. (2005) began measuring monetary shocks based on market instruments that have been extensively used since then, including the study of the reaction of expectations measures from other sources as, for example, Faust et al. (2004).

an informational advantage from the Federal Reserve.³ Then the literature was followed by Campbell et al. (2012) and Nakamura and Steinsson (2018), which present results for changes in inflation and unemployment or output expectations to a monetary surprise that not only take into account the surprise in the current policy rate but also the surprise in future rates. Using different methods, the former finds positive reactions to inflation forecasts (not very significant, but higher in number in the initial quarter ahead) and negative responses to unemployment forecasts (higher in the initial quarter ahead). In contrast, the latter finds no effect on two-year to 10-year inflation forecasts and a positive impact on output growth forecasts, especially in the first quarters. All these studies find positive coefficients, which contribute to the evidence of the information channel, but are based on monthly surveys and measures of the dependent and independent variables from different sources.

Since 2020, there was an increase in the monetary and information shocks of monetary decisions, focusing on disentangling both shocks and seeing if the "well-identified" monetary shock affects inflation and output with the correct theoretical signs. These include Jarociński and Karadi (2020), Miranda-Agrippino and Ricco (2021), Bu et al. (2021) and Nunes et al., 2023 among others. Among these, the ones that investigate the responses of expectations are Jarociński and Karadi (2020) and Nunes et al. (2023), which find evidence of positive reactions of one-year-ahead inflation and growth expectations to an information shock. Related to these studies, Güntner (2022) studied Blue Chip's forecasts revision to the information shock identified using the method in Jarociński and Karadi (2020) and found significant responses of output forecasts while non-significant of inflation forecasts. In all these cases, the expectations data is based on monthly surveys. The dependent and independent variables are not from the same source, and only Güntner (2022) includes the short term.

The only papers that investigate the revision of expectations to a monetary shock and that use inflation expectations measured at frequencies higher than one month are Andrade and Ferroni (2021) and Acosta (2023). The first uses market-based expectations starting one year ahead and measures the response to different monetary shocks (target, Delphic, and Odyssean). The closest to our shock is the target factor, which shows negative, non-significant responses of one-year inflation expectations. Compared to this study, the dependent and independent variables are measured from the same source; we

 $^{^{3}}$ D'Agostino and Whelan (2008) extended the analysis of Romer and Romer (2000), and found a deterioration of the informational advantage, but still superior forecast accuracy than professional forecasters for the very short term.

include shorter-term expectations, which is more beneficial for testing the information shock hypothesis, and our measurements are direct instead of including other variables that could be changing in response to the shock.

Acosta (2023), on the other hand, decomposes monetary policy announcements to shocks identified with the help of newspaper articles and shows the response of highfrequency, text-based measures of inflation and output expectations to those shocks. The results show positive responses of both expectations variables to perceived-demand shocks, negative in response to monetary shocks (current and news), positive growth response, and negative inflation response to a perceived supply shock. The measures of expectations are relatively short term and show an average of movements between the expectations of the same quarter and two quarters ahead. In comparison to this paper, the main differences are two-fold. First, they estimate different shocks based on the newspapers' reaction to the monetary announcement, and this paper estimates one shock, which is directly the unexpected component of the monetary decision since the interest is in whether the decision itself creates an informational effect. Secondly, they create high-frequency expectations measures, while the measures used in this paper are direct, so more accurate. In any case, the papers offer a different dimension to discussing monetary policy, the existence of different shocks, and their effects on expectations.

Specifically about Chile or Brazil, Aruoba et al. (2021) is the only paper that, with an identified monetary shock for Chile, shows the response of various economic variables, which include one-year-ahead inflation forecast using surveys. They show an initial positive response of inflation expectations. Among the differences with this paper is that the dependent and independent variables come from different sources, do not include shorterterm responses, and are not about revising expectations over some horizon but about the evolution of a fixed horizon.

The related theoretical literature includes papers with a signaling role to the monetary policy instrument. Melosi (2017) includes such a role in a dispersed information model to study the empirical relevance of the signaling effects of monetary policy for the United States and their implications for the spillover of policy and non-policy disturbances. In contrast to that model, the one presented in this paper assumes that all firms have the same information to focus on the conditions under which the model can explain the empirical results while keeping tractability. For this study, it is not necessary to go to dispersed information and introduce higher-order beliefs and the role of the interest rate as a coordination mechanism.⁴ If the mechanism under study is put in such an environment, the general conclusions will not be altered, but additional complications will be encountered. The model presented here is the baseline Neo-Keynesian model with a signaling role.⁵

The paper proceeds as follows. Section 2 briefly describes the monetary policy frameworks of the countries used in the empirical section and the specific datasets. It highlights the benefits of using that data and also the remaining concerns and what was done to deal with them. Section 3 presents the empirical analysis, including the main regressions and findings. It also shows the results of additional exercises and discusses some robustness exercises. Section 4 presents the model, describing the departures from the 3-equation baseline New-Keynesian model, the expectation formation processes, and the results about which shock can be behind the results found in the empirical section. Section 5 concludes.

2 Description of the Monetary Policy Frameworks and Datasets

As said in the introduction, the reason for using Chile and Brazil is because they meet three desirable conditions for studying the effects of monetary policy (including a possible information effect) on expectations. This section will present the main characteristics of the monetary policy frameworks of both countries and the main advantages and concerns about their datasets.

2.1 Monetary Policy Frameworks

Both countries are inflation targeters since 1999.⁶ Both set an overnight interest rate (the Selic rate in Brazil and the TPM in Chile) in order to achieve their inflation targets

 $^{^{4}}$ For theoretical literature about the role of public information and higher order beliefs Morris and Shin (2002) and (2003).

⁵Other papers include a signaling role in the monetary policy actions, but the instrument is not the interest rate. For example, Adam (2007), Tamura (2013), and Baeriswyl and Cornand (2010) study optimal monetary policy, Tang (2014) and Mertens (2011) study the effects that this signaling role has on discretionary policy and Berkelmans (2011) studies the effect of a signaling role on the propagation of shocks.

⁶Brazil had a formal change in June 1999, but only gained legal independence in 2021. Chile had the legal change including independence in 1989, but started with a crawling band until September 1999. Since then, a fully fledged inflation targeting regime has had a floating exchange rate.

set based on headline inflation measures of their main consumer price indexes.⁷ Both have operational independence, have a board that make their decisions by vote, currently meet 8 times per year (in normal times), among other common characteristics.

Both have operational independence, have a board that make their decisions by vote, currently meet eight times per year (in normal times), among other common characteristics. The main differences in their monetary policy frameworks since the periods of floating exchange rate regimes, is the setting of the targets, the general stability of the countries and their macroeconomic policies. Brazil sets inflation targets more than two years in advance, but that target moves around.⁸ In contrast, Chile has had a constant target centered at 3% since the flexible exchange rate period began.

Chile has been fairly stable for the whole period while Brazil has presented more instabilities. Carvalho and Nechio (2023) show and discuss that Brazil has experienced some episodes in which there were instabilities related to fiscal sustainability concerns and un-anchoring of inflation expectations.⁹ Chile on the other hand has experienced no critical change in its macroeconomic policy. The most important instability is the social unrest that erupted at the end of 2019, but did not have obvious macroeconomic policy repercussions.

2.2 Datasets

2.2.1 Brazilian Dataset

The analysis uses forecasts reported by the Market Expectations System (MES) of the Central Bank of Brazil, known as the Focus Survey. The MES was created in 1999 as part of the monetary policy framework and reports forecasts for several variables, including different inflation measures, output growth and the MPR, at different horizons. The survey is nowadays conducted online to professional forecasters and provides daily statistics of the responses in the system, so it is a high-frequency measure. The number of institutions is around 160, which undergo a pre-approval process and includes for example banks and asset managers.

⁷Selic stands for "Sistema Especial de Liquidação e Custódia" which means Special System for Settlement and Custody and TPM stands for "Tasa de Política Monetaria" which means Monetary Policy Rate.

 $^{^{8}}$ The first target was set in June 1999 and was 8% for the year 1999. The last target on record was decided in June 2023 and is 3% for the year 2026.

⁹Particularly high instabilities happen until end of 2002 and between the second-half of 2015 and August 2016 (Carvalho and Nechio, 2023; and references cited there).

The forecasts are not reported every day and are included in the statistics as long as they were reported in the last 30 days. There are incentives to update forecasts on specific dates because the MES publishes annual and monthly Top 5 rankings, which are closely followed by economic agents, making these rankings a significant factor in the financial landscape.

The sample used in the analysis is from the meetings held between November 2001 and December 2022, which implies 186 measurable meetings.¹⁰ The analyzed variables are monthly inflation expectations of the IPCA index, which is the main price index in Brazil, and of IP, and quarterly expectations of GDP.¹¹

2.2.2 Chilean Datasets

The datasets used for Chile are the EOF (Financial Traders Survey in Spanish) and the EEE (Economic Expectations Survey in Spanish). Both of them are professional forecasters' surveys conducted by the Centrak, Bank of Chile. The EOF is the preferred option, but it does not include output growth forecasts, which is the reason why a part of the analysis uses the EEE.

The EOF is less frequent than the Brazilian survey, but it has pre-meeting and an aftermeeting versions and includes questions about inflation and the MPR.¹² The available data includes the median and mean and the individual observations of each institution, which are around 50.¹³

Unlike the Brazilian survey, which is web-based and allows forecasters to enter data at their convenience, the Chilean survey is conducted each time, ensuring its complete up-to-dateness.

The CBC conducts another professional forecasters' survey, the EEE . This survey is used for the output expectations results. It is not our preferred option due to its monthly frequency and less up-to-date nature at the monetary policy meeting (its release date

¹⁰There was an extraordinary meeting in October 2002, which was excluded because of the lack of measure of the expected MPR.

¹¹The sample for IP stops in August 2021, because that forecast was discontinued.

¹²The release dates went from the 2nd and 4th Wednesdays of each month between 2009 and 2017 to 3 days before the press release of the meeting and 2 days after the release of the minute of the same meeting since 2018 (which happens around 2 weeks after the meeting). The reason behind this is that until 2017 the CBC had monthly meetings, starting in 2018 it had eight meetings every year.

¹³To secure the privacy of the individual agents, the Central Bank of Chile mandates that the development, extraction and publication of the results should not allow the identification, directly or indirectly, of natural or legal persons. Officials of the Central Bank of Chile processed the disaggregated data. All the analysis was implemented by the author and did not involve nor compromise the Central Bank of Chile.

depends on the release date of inflation), but it is the only option to study the reaction of output in Chile.

The sample for the analysis of inflation in Chile includes 137 meetings between January 2010 and October 2022. It analyzes the response of two monthly and two yearly inflation expectations. The sample for the GDP is between January 2002 and October 2022.

2.3 Benefits (and Concerns) of the Datasets

The main objective of the analysis is to measure responses of short-term output growth and inflation forecasts to surprises in the monetary policy decisions. To argue that a change in the monetary policy decision *causes* changes in expectations, one must control for many potential problems in doing this exercise. This section discusses the reasons behind the choice of the datasets, while highlighting remaining problems and the strategy to control them.

The reasons for choosing Brazil and Chile for this analysis, is because as far as my knowledge, their datasets are the only ones that meet three conditions: (i) they have "close-enough" observations before and after meetings; (ii) They include expectations at short term horizons; and (iii) they include expectations of the dependent and independent variable from the same source.

The first condition is important to avoid the occurrence of too many shocks between the measurements, which makes isolating the monetary shock very difficult. If frequency is low and the regressions lack enough controls, the results can show an information channel that is not there (as shown by Bauer and Swanson (2023)). This problem is not present in the Brazilian data because it has daily measures. In Chile, the frequency is longer, yet still very good in comparison with international standards.¹⁴ The concern using frequent surveys lies actually in allowing enough days for the survey measures to react.¹⁵

The second reason is the inclusion of short-term forecasts to connect with the information channel of monetary policy. Without an information channel, short-term expectations should not react, while medium-long term expectations should react negatively to a positive surprise. This is so because a higher-than-expected interest rate increases real rates, decreases current consumption, aggregate demand and inflation (and output). In

¹⁴For example, Blue Chip —the most common survey of professional forecasters cited in the literature— has only monthly measures.

¹⁵This contrast with market-based measures, as discussed by Nakamura and Steinsson (2018)), that react to many shocks even during one day. For discussion about the unresponsiveness of surveys refer to Coibion and Gorodnichenko (2012) and Coibion and Gorodnichenko (2015).

contrast, with an information channel, short-term expectations should react and mediumto-long term expectations are unclear, since the two shocks go in opposite directions. This condition contrasts with, for example, Andrade and Ferroni (2021) and Nakamura and Steinsson (2018) that report data only for longer horizons.

The last condition is that the surprise and the correction of the expectations come from the same source. This is important because different agents have different expectations and so when measuring from different sources they could be disconnected. The same decision can be a positive surprise for an agent (or group thereof) and a negative surprise for another, which would be especially problematic when connecting the shock and response of a variable, as is the case of this paper.

There are some issues from the use of these datasets that need to be recognized and that the analysis tries to control for. First, the monetary authorities do not surprise the market very often, which is shown in Figure 1. The figure shows in red the surprises measured using the median and in blue using the means. Using the medians, from a total of 186 and 131 measured surprises in Brazil and Chile, only 26% and 27% are non-zero, respectively. To control for this problem, the results shown in both countries use medians and means and in the case of Chile, also individual data.¹⁶

Second, and related to the previous, is that when measuring the expectations, especially using medians, not only are the surprises very few, but also the revision of the forecasts. This problem is related to the evidence that surveys are unresponsive (Coibion and Gorodnichenko, 2012) and is particularly problematic in Brazil, since there is no obligation to change the projections between measures. Figure 2 shows statistics about the relation between the surprises in the monetary policy decisions and the revision of inflation expectations. The figure shows the percentage of answers in the sample, per country and measure (medians/means/individual data) that have a positive correlation (blue), a negative correlation (red), both measures are zero (black), the surprise measure is zero, but the change of inflation expectations is not zero (dark gray), and the surprise is not zero, but the change in expectations is (light gray).

There are several interesting features that can be seen in the figure. First, there is a lot of black and dark gray using medians and individual data, which are the observations with no surprise in the policy rate, and so are uninformative. This problem is much

 $^{^{16}}$ Using means, 79% and 97% are non zero (75% if non-zero is defined as having an absolute value above 0.001). The percentages of non-zero observations are similar with individual data and medians. However, the availability of non-zero observations is much higher.



Notes: The graphs show the surprises available in the samples of each country. The surprises are measured as the difference between the decision and the mean forecast in blue and the median forecast in red. The top figure shows the case of Brazil and the bottom figure the case of Chile.

FIGURE 1: EVOLUTION OF SURPRISES: BRAZIL AND CHILE



Notes: The graphs show the percentage of responses in different categories depending on the surprise in the MPR and the revision of inflation expectations. Blue is when both go in the same direction; red is when they go in opposite directions; black is when both are zero; dark gray is when they revision of expectations is not zero, but the surprise is zero and light gray is when the surprise is not zero, but the revision of expectations is.

FIGURE 2: Relation Surprises and Revision of Inflation Expectations

smaller when using means in Brazil.¹⁷ In Chile, the problem is much less important using means and with individual data, even though there are big areas with those colors, the quantities are very large (over 4,000).

It is also important to highlight that using survey data with direct measures of expectations (inflation and monetary rates) avoids using approximate measures that can affect the results. On the one hand, when measuring inflation expectations using market-based measures, monetary shocks can affect the implicit expectation and other characteristics and, if not isolated properly, can contaminate the findings. On the other hand, when measuring the monetary surprise based on a rate different from the current policy rate itself (as in Nakamura and Steinsson (2018) or Jarociński and Karadi (2020) among others)), it has to be noted that this measure includes not only the surprise of the monetary policy decision itself, but also information given by the press release and short-term forward guidance. The shock measure here is then a lower bound of the information channel, since it excludes the information given (if any) by short-term forward guidance and also any endogenous reaction of expected short-term interest rate to a revision in expectations because of the monetary policy decision.¹⁸

3 Empirical Analysis

This section estimates the reaction of inflation and output growth forecasts in response to unexpected changes in the MPR. It presents the empirical strategy and the main results.

Even before going to the empirical strategy, Figure 2 shows the sign of the correlations between the surprises and the revision of expectations of the same month of the meeting and the second year after the date of the meeting in both countries. In most cases for the same month, the blue area is larger than the red area, meaning the positive correlation is more common than the negative one. In contrast, the areas are similar in the second year. Both observations contrast significantly with the prediction when there is only a monetary shock, which would be zero in the short term and negative in the long term.

¹⁷For Brazil, the differences in inflation expectations are taken waiting three days after the meeting. When waiting one, the dark-gray and black areas were even larger. See the Appendix for details.

¹⁸As in Nakamura and Steinsson (2018), Faust et al. (2004), Campbell et al. (2012) among others.

3.1 Main Regressions

The identification assumption is that during the period between the expectations measures before and after the meeting, the only shock that occurred was due to the monetary policy meeting. This assumption is natural for Brazil but more problematic for Chile. As a result, the exercises will include controls for Chile. The general regression of interest is:

$$z_{d+s,t+j}^{e} - z_{d-m,t+j}^{e} = \alpha_j + \beta_j (i_t - i_{d-m,t}^{e}) + C\left(_{\{d+s\}-\{d-m\}}\right) + \epsilon_{t,j}$$
(1)

where the meetings are held at day d of month t. The variables of the equation are the following: $z_{d+s,t+j}^{e}$ is the expectation of variable z j months in the future held s days after the meeting (the same way, $z_{d-m,t+j}^{e}$ is the expectation of the same variable for the same horizon, but m days before the meeting); i_t is the MPR decided by the Board at the meeting of day d and $i_{d-m,t}^{e}$ is the forecast of the MPR m days before the meeting. Finally, $C(_{\{d+s\}-\{d-m\}})$ are controls between the two dates that the dependent variable is measured. The coefficient of interest is β_h , which measures the effect that a surprise in the MPR decided at month t has on the change in the forecasts of z at month t + h.

The definitions of the specific variables and horizons are country-specific. For Brazil, z is inflation, IP, and GDP; m is zero because the initial expectations are from the same day of the meeting; and s = 3 because the expectations after the meeting are from three days after. In addition, j depends on each variable z and is $j = \{0, ..., 17\}$ for inflation, $j = \{-1, ..., 12\}$ for IP and $j = \{-1, 3\}$ for GDP.¹⁹ Because of the high frequency of data, these exercises exclude controls, $C = \emptyset$, but excludes dates that coincided with the release of the variable under study.

In the case of Chile, the primary variable under study is inflation (z), and the available horizons are $j = \{0, 1, 12, 24\}$, meaning the same month, one month ahead, one year ahead, and two years ahead. The measure before the meeting is, on average, done three days before, m = 3, and the measure after the meeting is, on average, done nine days after the meeting, s = 9. In addition, we include several controls that can affect macro variables: a measure of the surprise in inflation if the data was released (Surp. Infl.), change in the stock market index (Δ IPSA), change in the nominal exchange rate (Δ NER), change in the price of Copper (Δ Copp.P.), change in the price of oil (Δ WTI),

¹⁹In the case of *GDP*, the horizon is in quarters instead of months, so j changes accordingly. The appendix shows the results for $s = \{1, 2\}$.

change in the external price index (Δ IPE) and the last available output growth based on an index of economic activity (Δ Imacec).

Also in the case of Chile, there is a panel data version of (1) with horizon and individual-horizon fixed effects, and the measure of the expectations (both sides) depends on the institution *i* that answered the survey.

The main analysis also includes the reaction of expected GDP growth to the MPR surprise for Chile based on the Survey EEE. In that case z is GDP Growth, j is zero, because it is the same quarter of the meeting, and m and s vary more because the release of the survey is not fixed with respect to the meeting.

The regressions were estimated with OLS and robust standard errors to control for heteroskedasticity. For the panel, the results used clustered standard errors at the surveyed institution level.

3.2 Main Empirical Results

This section presents the reaction of inflation and output to surprises in the MPRs. It shows that both variables increase in the short run, and the response of the shorter run is higher than the response of the medium term.

3.2.1 The Response of Inflation Expectations

Figures 3 and 4 show the annualized coefficients of interest. The first columns show the results using medians, the second uses means, and the third (for Chile), uses individual data. The means and individual data graphs include a line with the results using medians for comparison. The Chilean results show the regressions using all the controls. For cases with fewer controls and the rest of the details, such as the rest of the coefficients and the number of observations, refer to Appendix A.2.



Notes: The graphs show the annualized response of inflation expectations given a change in the unexpected component of the MPR decision and their confidence intervals. Each graph shows the response of the current month (SM) until 17 months ahead (17M). The corresponding regressions are in Appendix A.2. The graph in b shows the coefficients of graph a in blue.

FIGURE 3: Brazil: Change in Inflation Forecasts

The figures show a positive short-term reaction of inflation expectations, given a positive surprise in the MPRs. This reaction is marginally significant for Brazil while very significant for Chile. The same graphs show that the reaction in the shorter terms is much stronger than in the medium term. The results using medians and means are comparable in both countries. In the case of Chile, the results based on the individual data are smaller than the ones using means and medians but highly significant.



Notes: The graphs show the annualized response of inflation expectations given a change in the unexpected component of the MPR decision (RPM in Chile) and their confidence intervals. Each graph shows the response of the following month, one month ahead, one year, and two years ahead. The graphs show the response of the regressions given all the controls. The regressions are in Appendix A.2. Graphs b. and c. show the coefficients of graph a. in red.

FIGURE 4: Chile: Change in Inflation Forecasts

The coefficients are also economically meaningful since the graphs show annualized inflation measures in percentages. Interpreting the coefficients, we have that in response to a 25-basis-point positive surprise in the MPRs, the forecast of the same month inflation increases by 0.17% - 0.25% in annualized terms for Brazil and 0.32% - 0.91% in the case of Chile.

Note also that the response to longer-term expectations is either non-significant or very small in magnitude. This result also contrasts with the case that the MPR only provides information on the monetary shock since one would expect inflation and output to decrease in the horizon that monetary policy affects the economy, which is between one and two years.

3.2.2 The Response of Output Expectations

Figure 5 and Table 1 show the results for output. In the case of Brazil, the results show that GDP increases for all available horizons, and IP shows an evolution similar to inflation, which increases more in the shorter term and less later. The first horizon of these cases is of particular importance since these are data that have already happened at the moment of the meeting but are not yet known. The importance of these relies on that, by definition, the monetary shock in these cases is non-existent, so there is only an information shock.

As with inflation, the Brazilian results using means and medians are similar in both cases and generally give the same message. In addition, it can be seen that both measures of output give a similar message in the first months, which is that output expectation increases in the short term and only differs slightly around one year in the future. GDP still shows a positive response, while IP decreases to zero or even slightly into negative territory.

The values are also economically significant. An unexpected increase of 25 basis points in the MPR implies an upward revision of around 0.01% in GDP and around 0.05% in IP.

The results for Chile, shown in Table 1, as discussed in section 2.2, are based on another survey (EEE), which is conducted monthly and asks only for the forecast of the same quarter of the survey. The results show positive coefficients in all the versions of the regressions, which is relatively stable in value, although it changes significance depending on the controls included.

The economic meaning in this case is that an unexpected increase of 25 basis points in the MPR increases GDP growth by around 0.15% in annualized terms.

To sum up, the results for short term inflation and output expectations show positive responses to a positive MPR surprise. These contrast with the interest rate, giving information only about the monetary shock and supporting the hypothesis of including an informational shock. In the medium term, the responses are mixed, which is more expected if there are informational and monetary shocks than only monetary shocks,

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Surp. MPR	0.626*	0.710**	0.709**	0.690**	0.683**	0.551^{*}	0.398	0.612
Surp. Infl.	(0.323)	$(0.331) \\ -0.262 \\ (0.250)$	$(0.328) \\ -0.261 \\ (0.248)$	$(0.335) \\ -0.270 \\ (0.240)$	$(0.335) \\ -0.280 \\ (0.245)$	$(0.326) \\ -0.297 \\ (0.256)$	(0.374) -0.297 (0.252)	$(0.463) \\ -0.256 \\ (0.241)$
Δ IPSA			$0.059 \\ (1.514)$	$0.404 \\ (1.305)$	$0.475 \\ (1.285)$	$1.022 \\ (1.353)$	$1.069 \\ (1.306)$	$1.359 \\ (1.301)$
Δ NER				$1.566 \\ (3.860)$	$0.950 \\ (4.160)$	$0.824 \\ (3.852)$	$0.487 \\ (3.673)$	$0.568 \\ (3.647)$
Δ Copp.P.					-0.579 (1.014)	1.053 (1.172)	$0.949 \\ (1.093)$	$0.787 \\ (1.143)$
Δ WTI						-2.694 (1.722)	-2.861^{*} (1.697)	-2.762 (1.786)
Δ IPE							0.145^{*} (0.078)	0.157^{**} (0.079)
Δ Imacec								-0.041 (0.038)
Constant	-0.008 (0.084)	0.001 (0.086)	0.000 (0.082)	-0.001 (0.085)	$0.003 \\ (0.087)$	$0.002 \\ (0.086)$	-0.035 (0.090)	$0.118 \\ (0.208)$
N Adj. R^2	$213 \\ 0.004$	$\begin{array}{c} 213 \\ 0.004 \end{array}$	213 -0.001	213 -0.004	213 -0.008	$\begin{array}{c} 213 \\ 0.034 \end{array}$	$213 \\ 0.052$	$\begin{array}{c} 213 \\ 0.066 \end{array}$
Standard errors in parentheses. * $p < 0.1$, ** $p < 0.05$, *** $p < 0.01$								

TABLE 1: Chile: Reaction Same Quarters' GDP, EEE, Median



Notes: The graphs show the response of GDP and IP expectations given a change in the unexpected component of the MPR decision and their confidence intervals. The GDP graphs show the previous quarter's (PQ) response until three quarters ahead (3Q). For IP, the graphs show the response from the previous month (PM) until 12 months ahead (12M). The regressions are in Appendix A.2. The graphs in b, show in blue the coefficients of graphs a. for GDP and IP, respectively.

FIGURE 5: Brazil: Change in Output Forecasts

because in these last cases those responses should be negative.

3.3 Additional Empirical Results

This section has two purposes. First, it allows for the possibility that the shock given by the unexpected component of the interest rate delivers different kinds of information depending on other characteristics happening at the same time. Second, it discusses some robustness exercises using different samples and additional measures.

The first set of exercises allows the effect of the MPR surprise to depend on the movement of the MPR and then to depend on the response of the stock market.

Finally, the section discusses some robustness exercises first checking if the results remain for small surprises and then using different samples for the analysis and different measures. For additional information about these exercises, refer to Appendix A.3.

This section reports only the results based on means and individual data to keep enough informative observations.

Dependence on Movements of the MPR

These results allow for the possibility that the surprise's effect on the expectations is

different if that surprise happened when the rate was not changed, decreased, or increased. The reason behind this is that in a world with imperfect information, agents could try to uncover information from different imperfect signals, including the change itself.

For example, if the interest rate is raised and a positive surprise occurs, both signals could be interpreted as information about high inflation (current or future). In contrast, if the same surprise occurs with no movement or with a reduction, the information given by the two signals goes in different directions, and so, expectations' responses could be different.

Figure 6 shows the results for inflation, GDP, and IP measured with means for Brazil and inflation measured with means and individual data for Chile. Three columns show the results, separating the cases with dummies. The first column presents the coefficient when there is no change in the interest rate, the second when the surprises and the movements go in the opposite direction (positive surprises with decreases and negative surprises with increases), and the last column when they go in the same direction (positive surprises with increases and negative surprises with decreases).

The results show that there are some changes, in general, in the expected direction. For Brazil, the biggest effect in inflation is when the movement and the surprise in the MPR go in the same direction. In contrast, when they go in opposite directions, the positive initial effect goes to negative territory, and when there is no change, it decreases in the first month. For output, there are mixed results. In the case of GDP, the effect increases when the surprise and the movement go in the same direction, is not much altered when they go in the opposite direction, and decreases without a movement in the interest rate. For IP, there is no apparent difference.

In the case of Chile, the analysis shows mixed results for inflation. As seen in Figure 6, the baseline results are very similar to the ones when the surprise and the movement go in the same direction. When they go in opposite directions, the initial positive coefficient disappears, the expectation for one month ahead slightly increases, and for further ahead decreases. When there is no change in the rate, shown in the first column of the figure, the results do not change much but lose significance and increase in the second month.

In short, the movement appears to be affecting the effect of the surprise in the majority of cases in the expected direction, but there is still an effect even without movement in the MPR.

Dependence on the Response of the Stock Markets

Some authors in the latest literature interpret a positive co-movement between the stock



Notes: The first three rows show the reaction of the Brazilian inflation, GDP, and IP expectations, measured with means. The last two rows show the reaction of Chilean inflation expectations measured with means ("mn") and individual data ("id"). The graphs separate the reaction when the surprise occurs without a change in the MPR (column a.), when there is a negative correlation between the surprise and the change (column b.), and when there is a positive correlation between the surprise and the change (column c.). The graphs show in blue the baseline results from Figures 3, 5 and 4. All the regressions for Chile include all the controls, and the ones with individual data have clusterized standard errors and individual fixed effects.

FIGURE 6: Dependence on the Movement of the MPR

market and the monetary surprise as an information shock, starting with Jarociński and Karadi (2020). This section presents the results when the regression separates the coefficients in cases with a positive and negative co-movement. It is important to note that this is an open debate, and other measures show no apparent relation, as in Bu et al. (2021).

Figure 7 shows the results for Brazil and Chile. As with Figure 6, the first three rows are the results for inflation, GDP, and IP measured with means for Brazil, and the last two are the results for inflation expectations measured with means and individual data for Chile. The column on the left shows the results when the surprise and the stock market negatively co-move, while the right column shows the positive case.

In Brazil, the reaction of inflation and GDP are effectively higher with a positive comovement, while in the case of IP, they are similar in the first month and lower in the following ones.

For Chile, the negative correlation results are similar to the main results without separation. In contrast, the initial month shows no update for the case with a positive co-movement, while one month ahead it is much higher and significant. The results for the last two versions, i.e., one year and two years ahead, are similar.

In sum, it is unclear whether the information shock happens when the stock market positively co-moves with the surprise. The results suggest that it is not necessary to have a positive co-movement to get these results. However, some variables differ when separating the reaction between a positive and negative co-movement with the stock market.

Additional Exercises

This subsection presents the main conclusions of several exercises that check for robustness and control by country-specific issues. Appendix A.3 shows more details.

There are three types of exercises: (i) testing for the dependence of the coefficients on small v/s big surprises, (ii) testing for the maintenance of the results in sub-samples, and (iii) testing using alternative inflation expectations measures.

The first exercise separates the coefficients between the cases where the surprises were "big" (greater than or equal to 50 basis points in absolute terms) and "small," which is the rest. The reason behind this exercise is to check if the results only happen when there are big surprises, or also in general. The findings show that in both countries, the coefficients for small surprises are higher and more significant than for large surprises, which leads to the conclusion that the channel is generally active, not only in extreme cases.



Notes: The first three rows show the reaction of the Brazilian inflation, GDP, and IP expectations, measured with means. The last two rows are the reactions to Chilean inflation expectations measured with means ("mn") and individual data ("id"). Column a. shows the result when the surprise coincides with the stock market's reaction in the opposite direction. Column b, shows when the surprise coincides with a movement in the Stock market in the same direction. The graphs show in blue the baseline results from Figures 3, 5, and 4. All the regressions for Chile include all the controls, and the ones with individual data have clusterized standard errors and individual fixed effects.

FIGURE 7: Dependence on the Reaction of the Stock Market

The second set of exercises investigates different sub-samples for each country. Brazil introduced the IT framework in 1999 and according to Carvalho and Minella (2009) and Cerisola and Gelos (2009) among others, there can be three periods defined: implementation and initial years (1999 until mid-2001), problematic period (mid-2001 until end of 2003) and consolidation (2004 onward).²⁰ In addition, there has been evidence of a period of un-anchored inflation expectations in Brazil between mid-2015 and September 2016. These sets of results then present the coefficients when excluding these unstable periods separately.

The results for the sub-samples in Brazil show a general decrease in the coefficient and significance of the effects on inflation in the first months. In contrast, there are no significant effects on GDP and IP expectations. If anything, they increase in general when excluding the instabilities.

For Chile, there are several exercises in this category. The first set of exercises excludes periods in which Chile was at the effective lower bound (ELB, between July 2009 and June 2010 and between April 2020 and July 2021). Secondly, exercises allowed different coefficients before and after 2018 because of changes in the regime. The changes included a decrease from twelve to eight meetings per year, more coordination between the meetings and the monetary report, and a lengthening of the press releases after every meeting. The last exercises include the possibility that the effect is different if the meeting occurs around the release of the Monetary Policy Report (IPoM). The amount of information given by the press release and the IPoM can be very different, and so can their effects, but many times they happen close enough, so their effect is difficult to separate.

Chile's results show no change in the signs or shape of the responses of any of the analyses mentioned above. The exercise that shows the most significant change is the differentiation between before and after 2018 and shows that the coefficients before 2018 were lower and less significant, while the ones after 2018 are close to the main results.

The last set of exercises shows the effects of a similar analysis but measures the change in expectations using market-based instruments. For Chile, the results show regressions using the one-day change in inflation expectations for different horizons using breakeven inflation measures and inflation insurance. Both cases show a relatively positive shortterm reaction, although they are less significant than the results based on surveys.

²⁰During the second period Brazil experienced many problems including a confidence crisis from the end of 2002 until the beginning of 2003 in which agents considered a regime change to a more controlled exchange rate likely. During this period, inflation forecasts from the MES changed their behavior. For additional information please refer to Bevilaqua et al. (2008), Minella et al. (2003), Cerisola and Gelos (2009).

In sum, in some cases the significance of the results depends on the sample, but not their signs. That is expected and is part because there are fewer observations and also because there could be changes in the regimes that can alter the sensitivity of the effect of MPR surprises.

4 Simple Model with an Information Shock

This section presents the simplest possible model in which we can allow for the interest rate decision made by the central bank to cause expectations to react. We will maintain the assumption of rationality, and allow a temporary asymmetry of information, which resolves before making consumption and pricing decisions.²¹ The model is FIRE (Full Information Rational Expectations), but it is very explicit on how and which kind of information agents receive from seeing the interest rate.

The final equations and equilibrium of the model are the same as the log-linearized baseline three equations Neo-Keynesian model, as in Gertler et al. (1999), Woodford (2003, Chapter 4), Galí (2008, Chapter 3) and Walsh (2010, Chapter 8).

The first part presents the final equations and equilibrium of the model; it then explains the temporary asymmetry of information and the timing when expectations of inflation and output are measured. It finally shows how expectations react depending on the type of information given by the interest rate.

4.1 Log-linearized Equilibrium Conditions

This subsection presents the log-linearized equilibrium conditions of the model to explain the potential sources of information given by the interest rate. Appendix A.4 presents the problems of each agent, the first-order conditions, and the equilibrium equations. The 3-equation log-linearized equilibrium conditions are:

$$\pi_t = \kappa x_t + \beta \mathbb{E}_t \pi_{t+1} + u_t \tag{2}$$

$$x_{t} = \mathbb{E}_{t} x_{t+1} - \sigma(i_{t} - \mathbb{E}_{t} \pi_{t+1} - r_{t}^{*})$$
(3)

$$i_t = \phi_\pi \pi_t + \phi_x x_t + \epsilon_t \tag{4}$$

 $^{^{21}\}mathrm{This}$ is not necessary to get the results, but it simplifies the equations and does not affect the sign of the results.

where the endogenous variables are π_t , x_t , and i_t , which are inflation, the output gap, and the interest rate, respectively. The first equation is the Phillips curve, which comes from the optimality conditions of the firms. The second equation is the IS relation, which comes from the households' Euler equation, and the final equation is the Taylor rule, which describes the monetary policy.

Three exogenous processes or shocks in these equations are the source of variations in the endogenous variables. The first is the cost-push shock, u_t , which is in the Phillips curve equation and arises due to variations in the elasticity between the varieties of goods that become the final consumption good. The second is the natural rate of interest, r_t^* , arising from variations in government expenditure and productivity. Finally, a monetary shock, ϵ_t , arises as the exogenous component of the Taylor rule.

The rational expectations equilibrium is defined by the processes $\{\pi_t, x_t, i_t\}$ consistent with (2)-(4) given the exogenous shocks $\{u_t, r_t^*, \epsilon_t\}$ and their distributions. These equations, written in terms of a 2x2 system, are:

$$\mathbb{E}_t z_{t+1} = A z_t + \xi_t \tag{5}$$

with

$$z_t = \begin{bmatrix} \pi_t \\ x_t \end{bmatrix} \quad \xi_t = \begin{bmatrix} -\frac{1}{\beta}u_t \\ \sigma(\epsilon_t - r_t^*) + \frac{\sigma}{\beta}u_t \end{bmatrix} \quad A = \begin{bmatrix} \frac{1}{\beta} & -\frac{\kappa}{\beta} \\ \sigma\left(\phi_{\pi} - \frac{1}{\beta}\right) & 1 + \sigma\left(\phi_x + \frac{\kappa}{\beta}\right) \end{bmatrix}$$

Assuming the Taylor principle, $\phi_{\pi} + \phi_x(1-\beta)/\kappa > 1$, the unique bounded rational expectations solution is:

$$z_t = -\sum_{j=0}^{\infty} A^{-(j+1)} \mathbb{E}_t \xi_{t+j}$$
(6)

This shows that the endogenous variables, including the interest rate, are a function of current and expected future exogenous shocks.

4.2 Timeline and Expectations Formation

This section analyzes the signs of the responses of inflation and output expectations when the interest rate gives information of different shocks. To do that, the model assumes a temporary information asymmetry between the monetary authority that sets the



Notes: CB = Central Bank.



interest rate and the rest of the agents. Before the agents make their final decisions, the information becomes perfect for all (hence symmetric).

As shown in Figure 8, the timeline is as follows. At the beginning of the period, agents can get partial information and set expectations of any variable n before seeing the interest rate, $n_{t,d}^e$. Then, the central bank gets information on the shocks hitting the economy in period t, and with that information, it sets the interest rate, which everybody sees. After seeing the interest rate, agents set new expectations, $n_{t,d+1}^e$. Finally, agents get the rest of the information not given by the interest rate (if any) and make their pricing and consumption decisions.

The model presented here is the simplest model in which the interest rate provides information. Expectations are still rational and constrained only temporarily by pending information but are common for all. The model excludes many departures from these assumptions that could also affect the results, such as imperfect information (common or dispersed) from the firm side, or from the household side or both.²² In all cases, though, the intuition behind the results shown in this section remains.

For simplicity, it will be assumed that the three shocks are *i.i.d.* normally distributed, $\omega_t \sim N(0, \sigma_{\omega}^2)$, with $\omega_t = \{\epsilon_t, u_t, r_t^*\}$. Additionally, to allow the interest rate to give information about future events in the economy, the model includes news shocks in the form of a noisy signal about a future shock. The news shocks in t are the following: $s_t^{\omega} \sim N(\omega_{t+1}, \sigma_{s_{\omega}}^2)$, with $\omega = \{\epsilon, u, r^*\}$.

Solving equation (6) (with equation (4) to solve for the interest rate) and the shocks described in the previous paragraph gives solutions of the form:

$$\omega_t = \Lambda^{\omega}_{\epsilon} \epsilon_t + \Lambda^{\omega}_u u_t + \Lambda^{\omega}_r r^*_t + \Lambda^{\omega}_{s^{\epsilon}} s^{\epsilon}_t + \Lambda^{\omega}_{s^u} s^u_t + \Lambda^{\omega}_{s^{r^*}} s^{r^*}_t$$
(7)

 $^{^{22}}$ For models with dispersed information from the firm side see Melosi (2017) and from the household side see Wiederholt (2015).

for $\omega_t = \{\pi_t, x_t, i_t\}$, with Λ_{ξ}^{ω} function of the parameters in equation (3)-(5) for $\omega_t = \{\pi_t, x_t, i_t\}$ and $\xi = \{\epsilon, u, r_t^*, s^{\epsilon}, s^{u}, s^{r^*}\}$. With equation (7), we can calculate the expectation of any of the variables at day d as:

$$\omega_{t,d}^e = \Lambda_{\epsilon}^{\omega} \epsilon_{t,d}^e + \Lambda_u^{\omega} u_{t,d}^e + \Lambda_r^{\omega} r_{t,d}^{*,e} + \Lambda_{s^{\epsilon}}^{\omega} s_{t,d}^{e,e} + \Lambda_{s^{u}}^{\omega} s_{t,d}^{u,e} + \Lambda_{s^{r^*}}^{\omega} s_{t,d}^{r^*,e}$$
(8)

with $y_{t,d}^e$ the expectations of variable/shock y_t held at day d for any y. This equation shows that the interest rate will affect inflation and output expectations if it gives information about any of these shocks that was not available before. By also using this equation we can calculate the revision in expectations of any variables between day d and d+1, which are defined as the dates before seeing the interest rate and after seeing the interest rate respectively. Note that the expectation of the rate will be the rate itself at d+1, which means $i_{t,d+1}^e = i_t$. The expectation and realization of inflation and output though, could perfectly still differ in d+1 if with the information provided by the interest rate there remains some asymmetry of information.

With all this in hand, we can get the coefficients of interest which are the ones that relate the surprise in the interest rate with the revision in inflation and output expectations, which are χ_{π} and χ_{x} in the following relations:

$$\pi_{t,d+1}^e - \pi_{t,d}^e = \chi_{\pi}(i_t - i_{t,d}^e) \tag{9}$$

$$x_{t,d+1}^e - x_{t,d}^e = \chi_x(i_t - i_{t,d}^e)$$
(10)

Note that all terms concerning shocks that agents know perfectly (get that info in the first arrow of the timeline shown in Figure 8) before the central bank sets the interest rate are going to be canceled out on both sides of equations (9) and (10). Only new relevant information is going to remain.

It is essential to highlight that these agents have a lot of information about the economy. They know with certainty the equations of the economy, how the central bank sets the interest rate, and which shocks can hit.

4.3 Results of the Model

This subsection will present qualitative results about the coefficients in (9) and (10) when the interest rate gives information about different shocks. In the first part, the interest gives information about one shock each time, while in the second part, it gives

	$C\iota$	urrent Shoo	cks	News shocks			
	ϵ_t	u_t	r_t^*	s_t^ϵ	s^u_t	$s_t^{r^*}$	
χ_{π}	_	+	+	+	+	+	
χ_x	_	_	+	+	_	+	

TABLE 2: The Interest Rate Gives Information about One Shock

information about two shocks: the current monetary shock and an additional one.

The exact expression of the coefficients depends on the information provided by the interest rate. Take for example the case that the interest rate gives information only about the current monetary shock, ϵ_t . In that case, since it is one signal about one unknown, the agents get full information after seeing the interest rate and the differences in expectations are going to be the following:

$$\pi^{e}_{t,d+1} - \pi^{e}_{t,d} = \Lambda^{\pi}_{\epsilon}(\epsilon_{t} - \epsilon^{e}_{t,d})$$
$$x^{e}_{t,d+1} - x^{e}_{t,d} = \Lambda^{x}_{\epsilon}(\epsilon_{t} - \epsilon^{e}_{t,d})$$
$$i_{t} - i^{e}_{t,d} = \Lambda^{i}_{\epsilon}(\epsilon_{t} - \epsilon^{e}_{t,d})$$

leading to the coefficients of interest:

$$\chi_{\pi} = \frac{\Lambda_{\epsilon}^{\pi}}{\Lambda_{\epsilon}^{i}} < 0 \qquad \qquad \chi_{x} = \frac{\Lambda_{\epsilon}^{x}}{\Lambda_{\epsilon}^{i}} < 0$$

Appendix A.4 shows the specification of the coefficients of interest and demonstrate that they are both negative, which is the information given in the first column of Table 2. The rest of the table shows the signs of the coefficients χ_{π} and χ_x when the interest rate gives information about one shock each time, for each current and news shock. These exercises imply that if, for example, the interest rate gives information about the news of the monetary shock (first column of *News shocks*), the agents can see all shocks except that one since the beginning of the period perfectly.²³

The results for current shocks show that inflation and output expectations respond positively only to the natural rate of interest shock. The intuition behind the results for

Notes: All results are for parameter values that satisfy the Taylor Principle $(\phi_{\pi} + \phi_x(1-\beta)/\kappa > 1)$. For the expressions in each case, refer to Appendix A.4. The results of the news shocks depend on parameter values. The Appendix also reports the parameter values.

²³There can be two interpretations about this: either all the shocks except the one under study are perfect information for all agents since the beginning of the period, or all the other shocks do not exist (except the shock about which news exists in the case of news shocks).

each shock is straightforward. First, both expectations react negatively to information about the monetary shock since it increases the short-term real rate, decreases current demand, and causes a drop in inflation and output.

In contrast, when the information is about the cost-push shock, there is an upward revision of inflation expectations and downward of output expectations since this shock causes an increase in the desired markup, which raises inflation and contracts output. Only the natural interest rate causes a positive update in both expectations because that shock causes an increase in both variables and an increase in the nominal interest rate as a result of the endogenous reaction of inflation and output.²⁴

When the interest rate gives information about news, shown in the right columns of Table 2, the signs about monetary news are reversed compared to the current monetary policy shock. The reason is that while a positive monetary policy shock tomorrow (and hence news about it) decreases inflation and output today, as with current monetary policy shocks, it also decreases the interest rate today.

In the case of news about the natural rate, current inflation, output and interest rate increase as a reaction. The effects of positive future monetary shocks in this model are the same as a negative future natural rate of interest. Finally, when the interest rate gives information about future cost-push shocks, the results show the same signs as in the case of current cost-push shocks.

To sum up, if the interest rate gives information about one shock each time, the information has to be about the natural rate and about news of the future monetary shock and future natural rate to two positive coefficients.

The limitation of the results shown in the previous version is that the interest rate can only give information about one shock each time. When shocks other than monetary shocks were analyzed, the assumption was that agents had received information about the shock from another source or that this shock did not exist. While it is likely that agents have information about the cost-push shock and the natural rate of interest from other sources, they are unlikely to have a source other than the MPR about the monetary shock. A more realistic assumption is that the interest rate gives information about the current monetary shock and something else.

Table 3 shows the results when the interest rate gives information about the current monetary shock, ϵ_t , and an additional shock in each column. The first column reports the result for the natural rate, r_t^* , the second about news of the monetary shock, s_t^{ϵ} , and the

 $^{^{24}\}mbox{For example, an increase in current government expenditure, which increases demand and prices and output.$

TABLE 3: The Interest Rate Gives Information about Two Shocks



Notes: All results are for parameter values that satisfy the Taylor Principle $(\phi_{\pi} + \phi_x(1-\beta)/\kappa > 1)$. Refer to Appendix A.4 for the exact expressions. The results of the last two columns are numerical for the ranges of parameters described in the Appendix.

last one about news of the natural rate, $s_t^{r^*}$. The exercise excludes current and news about the cost-push shock since the combination with the monetary shock can never provide a positive answer to the output gap (shown by the negative coefficients in that response in table 2).

The table shows that positive coefficients become a possibility when the interest rate provides information about the monetary shock and the natural rate and about the monetary shock and news of the natural rate. In contrast, the output response is never positive when the interest rate gives information about current and future monetary shocks. This result is because in that case the information provided about the current shock always more than outweighs the information about the future shock.

In contrast, when the interest rate provides information about the monetary shock and the current or future natural rate, the responses can be positive as long as the latter's information is "more important" than the former's. Moreover, this happens when the variability of the natural rate is high enough compared to the variability of the monetary shock. And the intuition is as follows: when the agent sees a surprise in the interest rate, it will attribute more of that surprise to a change of the shock that is more variable.

These findings, while acknowledging the potential influence of other factors such as private information of the agents, provide valuable insights. They suggest that, on average, in the economy, the responses of inflation and output expectations to a surprise movement in the interest rate can be interpreted as agents perceiving the interest rate as a source of information about a demand shock. This interpretation, encapsulated in the model as the natural rate or news about the natural rate, has significant implications for our understanding of macroeconomic dynamics.

5 Conclusion

This paper presented evidence on the responses of inflation and output growth forecasts to unexpected increases in the MPR. The analysis reported results based on the best possible datasets that met three conditions: (i) they provide close "before meeting" and "after meeting" measures (ii) they include expectations at short-term horizons; and (iii) they include expectations of the dependent and independent variable from the same source.

The results show that, in response to an unexpected increase in the MPR, professional forecasters revise their inflation and output forecasts upward in the short run. For the medium term, the effects are near zero.

This is a surprising result for current macroeconomic theory since surprises in the interest rate set by the central bank are associated with monetary shocks, which have negative effects on inflation and output. So, if rational, the responses of current inflation and output growth forecasts should be negative.

A baseline New-Keynesian model can explain the results if it allows the interest rate to provide information about shocks other than the monetary shock. It was shown that in order to obtain positive responses to inflation and output forecasts, the interest rate must provide "enough" information about the current natural rate of interest or news about the future natural rate of interest.

This analysis focuses only on the average reaction of expectations to the surprise MPR decision. It has to be highlighted that there are other types of communication at the time of the decisions, which are also present and could also affect expectations. This is for example the case of the information provided by the press releases, in addition to the decision, which is also present in Brazil and Chile among other countries as shown by González and Cruz Tadle (2021). The information given by the rest of the press release could be an additional information channel of monetary policy and separate the information given by the decision itself and by the rest of the release. This is a matter for future research.

References

- Acosta, Miguel (2023). "The Perceived Causes of Monetary Policy Surprises".
- Adam, Klaus (2007). "Optimal monetary policy with imperfect common knowledge". Journal of Monetary Economics 54 (2), pp. 267–301.
- Andrade, Philippe and Filippo Ferroni (2021). "Delphic and Odyssean Monetary Policy Shocks: Evidence from the Euro Area". Journal of Monetary Economics 117, pp. 816– 832.
- Aruoba, Boragan, Andrés Fernández, Daniel Guzmán, Ernesto Pastén, and Felipe Saffie (2021). "Monetary Policy Surprises in Chile: Measurement and Real Effects". Working Paper. N°921.
- Baeriswyl, Romain and Camille Cornand (2010). "The Signaling Role of Policy Actions". Journal of Monetary Economics 57 (6), pp. 682–695.
- Bauer, Michael D. and Eric T. Swanson (2023). "An Alternative Explanation for the Fed Information Effect". American Economic Review 113 (3), pp. 664–700.
- Berkelmans, Leon (2011). "Imperfect information, multiple shocks, and policy's signaling role". *Journal of Monetary Economics* 58 (4), pp. 373–386.
- Bevilaqua, Afonso S., Mário Mesquita, and André Minella (2008). "Brazil: Taming Inflation Expectations". Transmission Mechanisms for Monetary Policy in Emerging Market Economies. Ed. by Bank for International Settlements. Vol. 35. Bank for International Settlements, pp. 139–158.
- Bu, Chunya, John Rogers, and Wenbin Wu (2021). "A Unified Measure of Fed Monetary Policy Shocks". Journal of Monetary Economics 118, pp. 331–349.
- Calvo, Guillermo A. (1983). "Staggered Prices in a Utility-Maximizing Framework". Journal of Monetary Economics 12 (3), pp. 383–398.
- Campbell, Jeffrey R., Charles L. Evans, Jonas D.M. Fisher, and Alejandro Justiniano (2012). "Macroeconomic Effects of Federal Reserve Forward Guidance". Brookings Papers on Economic Activity 44 (1), pp. 1–80.
- Carvalho, Carlos and Fernanda Nechio (2023). "Challenges to Disinflation: The Brazilian Experience". 54 (1), pp. 217–241.
- Carvalho, Fabia and André Minella (2009). "Market Forecasts in Brazil: Performance and Determinants". Working Papers Series. Working Papers Series, Central Bank of Brazil, April.
- Cerisola, Martin and R. Gaston Gelos (2009). "What Drives Inflation Expectations in Brazil? An Empirical Analysis". *Applied Economics* 41 (10), pp. 1215–1227.
- Cieslak, Anna and Andreas Schrimpf (2019). "Non-monetary news in central bank communication". Journal of International Economics 118, pp. 293–315.
- Cochrane, John (2015). "Doctrines Overturned". *The Grumpy Economist*, February 28, [URL: http://johnhcochrane.blogspot.com/2015/02/doctrines-overturned.html].
- Coibion, Olivier and Yuriy Gorodnichenko (2012). "What Can Survey Forecasts Tell Us about Information Rigidities?" Journal of Political Economy 120 (1), pp. 116–159.
- (2015). "Information Rigidity and the Expectations Formation Process: A Simple Framework and New Facts". American Economic Review 105 (8), pp. 2644–78.
- D'Agostino, Antonello and Karl Whelan (2008). "Federal Reserve Information During the Great Moderation". Journal of the European Economic Association 6 (2-3), pp. 609– 620.
- Faust, Jon, Eric Swanson, and Jonathan H. Wright (2004). "Do Federal Reserve Policy Surprises Reveal Superior Information about the Economy?" The B.E. Journal of Macroeconomics 4 (1), pp. 1–31.
- Galí, Jordi (2008). Monetary Policy, Inflation and the Business Cycle: An Introduction to the New Keynesian Framework. Princeton University Press.
- García-Schmidt, Mariana and Michael Woodford (2019). "Are Low Interest Rates Deflationary? A Paradox of Perfect-Foresight Analysis". American Economic Review 109 (1), pp. 86–120.
- Gertler, Mark, Jordi Galí, and Richard Clarida (1999). "The Science of Monetary Policy: A New Keynesian Perspective". *Journal of Economic Literature* 37 (4), pp. 1661–1707.
- González, Mario and Raul Cruz Tadle (2021). "Monetary Policy Press Releases: An International Comparison". Working Paper. N°912.
- Güntner, Jochen (2022). "Central Bank Information and Private-Sector Expectations". Journal of Forecasting 41 (7), pp. 1372–1385.
- Gürkaynak, Refet S., Brian Sack, and Eric Swanson (2005). "o Actions Speak Louder Than Words? The Response of Asset Prices to Monetary Policy Actions and Statements". International Journal of Central Banking 1 (1), pp. 55–93.
- Hoesch, Lukas, Barbara Rossi, and Tatevik Sekhposyan (July 2023). "Has the Information Channel of Monetary Policy Disappeared? Revisiting the Empirical Evidence". *American Economic Journal: Macroeconomics* 15 (3), pp. 355–87.
- Jarociński, Marek and Peter Karadi (2020). "Deconstructing Monetary Policy Surprises—The Role of Information Shocks". American Economic Journal: Macroeconomics 12 (2), pp. 1–43.

- Kuttner, Kenneth N. (2001). "Monetary Policy Surprises and Interest Rates: Evidence from the Fed Funds Futures Market". Journal of Monetary Economics 47 (3), pp. 523– 544.
- Melosi, Leonardo (2017). "Signaling Effects of Monetary Policy". *Review of Economic Studies* 84 (2), pp. 853–884.
- Mertens, Elmar (2011). "Managing Beliefs about Monetary Policy under Discretion". Discussion Paper, Board of Governors of the Federal Reserve System, August.
- Minella, André, Paulo Springer de Freitas, Ilan Goldfajn, and Marcelo Kfoury Muinhos (2003). "Inflation Targeting in Brazil: Constructing Credibility under Exchange Rate Volatility". Journal of International Money and Finance 22 (7), pp. 1015–1040.
- Miranda-Agrippino, Silvia and Giovanni Ricco (July 2021). "The Transmission of Monetary Policy Shocks". American Economic Journal: Macroeconomics 13 (3), pp. 74– 107.
- Morris, Stephen and Hyun Song Shin (2002). "Social Value of Public Information". American Economic Review 92 (5), pp. 1521–1534.
- (2003). "Global Games Theory and Applications". Advances in Economics and Econometrics, 8th World Congress of the Econometric Society. Ed. by M. Dewatripont, L. Hansen, and S. Turnovsky. Cambridge University Press.
- Nakamura, Emi and Jón Steinsson (2018). "High Frequency Identification of Monetary Non-Neutrality: The Information Effect". Quarterly Journal of Economics 133 (3), pp. 1933–1980.
- Nunes, Ricardo, Ali Ozdagli, and Jenny Tang (2023). "Interest Rate Surprises: A Tale of Two Shocks". School of Economics Discussion Papers 0923. Working Paper, Federal Reserve Bank of Chicago, revised July.
- Romer, Christina and David Romer (2000). "Federal Reserve Information and the Behavior of Interest Rates". *American Economic Review* 90 (3), pp. 429–457.
- Smith, Noah (2014). "There's More to QE than Krugman Thinks". Bloomberg View, November 11 [URL: http://www.bloombergview.com/articles/2014-11- 11/theresmore-to-qe-than-krugman-thinks].
- Tamura, Wataru (2013). "Optimal Monetary Policy and Transparency under Informational Frictions". Working Paper. Unpublished, October.
- Tang, Jenny (2014). "Uncertainty and the Signaling Channel of Monetary Policy". Working Paper. Working Paper, Federal Reserve Bank of Boston, April.

- Varela, Felipe R. (2007). "Mercados de Derivados: Swap de Tasas Promedio Cámara y Seguros de Inflación". Studies in Economic Statistics. N°56, April.
- Walsh, Carl (2010). Monetary Theory and Policy, Third Edition. 3rd ed. Vol. 1. The MIT Press.
- Wiederholt, Mirko (2015). "Empirical Properties of Inflation Expectations and the Zero Lower Bound". Working Paper. Unpublished, June.
- Williamson, Stephen (2014). "Neo-Fisherians: Unite and Throw Off MV = PY and Your Phillips Curves!" New Monetarist Economics, November 13, [URL: http://http://newmonetarism.blogspot.com/2014/11/neo-fisherians-unite-and-throwoff-mvpy.html].
- Woodford, M. (2003). Interest and Prices: Foundations of a Theory of Monetary Policy. Princeton University Press.

Appendix



A.1 Additional Descriptive Statistics

Notes: The graphs show the percentage of responses in different categories depending on the surprise in the MPR and the revision of GDP/IP expectations using medians and means and for different horizons. Blue indicates when both go in the same direction; red indicates when they go in opposite directions; black indicates when both are zero; dark gray indicates when the revision of expectations is not zero, but the surprise is zero; and light gray indicates when the surprise is not zero, but the revision of expectations is.

FIGURE 9: Characteristics of Output Samples in Brazil



Notes: The graphs show the percentage of responses in different categories depending on the surprise in the MPR and the revision of IPCA expectations when waiting one, two, and three days. Blue is when both go in the same direction; red is when they go in opposite directions; black is when both are zero; dark gray is when the revision of expectations is not zero, but the surprise is zero, and light gray is when the surprise is not zero, but the revision of expectations is.

FIGURE 10: Characteristics of IPCA per day

A.2 Appendix to the Main Empirical Results

A.2.1 Brazil

This appendix shows the tables with all the regressions for Brazil presented in the main part of the paper. These regressions are based on waiting 3 days for the expectations to react. In addition, the section also shows graphs when allowing 1 and 2 days in all the measures.

	Surp.	MPR	Cons	tant	N	R^2
SM	0.092	(0.07)	0.005	(0.003)	138	0.132
$1\mathrm{M}$	0.03	(0.022)	0.005***	(0.002)	138	0.071
2M	0.005	(0.004)	-0.001	(0.001)	138	0.013
3M	-0.001	(0.007)	-0.002*	(0.001)	138	0
4M	-0.001	(0.003)	0.001	(0.001)	138	0.001
$5\mathrm{M}$	-0.002	(0.003)	0.001	(0.001)	138	0.003
6M	-0.006	(0.006)	0	(0.001)	138	0.016
7M	-0.001	(0.003)	0	(0.001)	138	0.001
8M	-0.001	(0.002)	0	(0.001)	138	0.001
9M	-0.01*	(0.005)	0	(0.001)	138	0.08
10M	-0.001	(0.003)	0	(0.001)	138	0
11M	0.003	(0.006)	0	(0.001)	138	0.004
12M	0.001	(0.002)	0.002***	(0.001)	138	0.001
13M	0.003	(0.01)	0.002	(0.002)	135	0.001
14M	-0.003	(0.003)	0	(0.001)	77	0.017
15M	-0.003	(0.003)	0	(0.001)	77	0.013
16M	-0.005	(0.005)	0	(0.001)	77	0.024
17M	0.017	(0.017)	-0.001	(0.002)	71	0.051

TABLE 4: Brazil: Reaction of Inflation Forecasts, Medians

	Surp.	MPR	Cons	stant	N	R^2
SM	0.06	(0.051)	0.006^{*}	(0.003)	138	0.062
$1\mathrm{M}$	0.023	(0.018)	0.004^{**}	(0.002)	138	0.039
2M	0.015^{**}	(0.007)	0	(0.001)	138	0.057
3M	0.004	(0.007)	-0.001	(0.001)	138	0.002
4M	-0.001	(0.004)	-0.001	(0.001)	138	0.001
$5\mathrm{M}$	-0.001	(0.003)	0.001	(0.001)	138	0.001
6M	0	(0.003)	0	(0)	138	0
7M	0.003	(0.002)	0	(0)	138	0.008
8M	-0.003	(0.004)	0.001	(0.001)	138	0.006
9M	-0.002	(0.003)	-0.001	(0)	138	0.005
10M	-0.003	(0.002)	-0.001	(0)	138	0.02
11M	-0.003	(0.004)	0	(0)	138	0.016
12M	0.001	(0.005)	0	(0.001)	138	0.001
13M	0.001	(0.006)	0	(0.001)	135	0
14M	-0.003	(0.003)	0	(0.001)	77	0.014
15M	-0.003	(0.004)	0	(0.001)	77	0.007
16M	-0.005	(0.004)	0	(0.001)	77	0.02
17M	0.016	(0.013)	0	(0.001)	71	0.066

TABLE 5: Brazil: Reaction of Inflation Forecasts, Means

TABLE 6: Brazil: Reaction of GDP Forecasts, Medians

	Surp.	MPR	Const	tant	Ν	R^2
PQ	0.034	(0.024)	-0.003	(0.006)	112	0.013
SQ	0.065^{*}	(0.036)	-0.021***	(0.007)	169	0.023
1Q	0.033	(0.021)	-0.018***	(0.007)	169	0.006
2Q	0.059	(0.038)	-0.025***	(0.007)	169	0.022
3Q	0.045	(0.039)	-0.029***	(0.006)	169	0.014

TABLE 7: Brazil: Reaction of GDP Forecasts, Means

	Surp.	MPR	Const	tant	N	\mathbb{R}^2
\mathbf{PQ}	0.035**	(0.016)	-0.004	(0.005)	112	0.023
SQ	0.054	(0.033)	-0.017**	(0.007)	169	0.019
1Q	0.092	(0.062)	-0.03**	(0.012)	169	0.019
2Q	0.041	(0.043)	-0.026***	(0.007)	169	0.011
3Q	0.036	(0.031)	-0.018***	(0.006)	169	0.011

TABLE 8: Brazil: Reaction of IP Forecasts, Medians

	Surp.	MPR	Const	tant	N	R^2
РМ	0.214	(0.14)	-0.057	(0.064)	160	0.004
SM	0.159^{*}	(0.092)	-0.101***	(0.033)	163	0.008
$1\mathrm{M}$	0.275^{**}	(0.134)	-0.06**	(0.029)	163	0.03
2M	0.195	(0.134)	-0.094***	(0.028)	163	0.017
3M	0.182	(0.133)	-0.047	(0.028)	163	0.014
4M	0.187	(0.131)	-0.028	(0.036)	163	0.009
$5\mathrm{M}$	0.121	(0.087)	-0.009	(0.026)	163	0.008
6M	0.103	(0.091)	-0.009	(0.026)	163	0.005
7M	0.158	(0.149)	-0.009	(0.021)	163	0.018
8M	0.047	(0.103)	-0.097*	(0.054)	163	0
9M	0.074	(0.181)	-0.037	(0.038)	163	0.001
10M	0.12	(0.089)	0.007	(0.053)	163	0.002
11M	-0.077	(0.092)	0.006	(0.034)	163	0.002
12M	-0.187	(0.209)	0.015	(0.032)	157	0.014

	Surp.	MPR	Cons	tant	N	\mathbb{R}^2
PM	0.225^{**}	(0.101)	-0.105***	(0.039)	146	0.016
SM	0.093^{*}	(0.055)	-0.037	(0.025)	163	0.005
1M	0.099	(0.081)	-0.052**	(0.025)	163	0.006
2M	0.08	(0.079)	-0.004	(0.024)	163	0.004
3M	0.039	(0.061)	0	(0.02)	163	0.001
4M	0.066	(0.061)	-0.03	(0.02)	163	0.004
5M	0.025	(0.07)	-0.007	(0.024)	163	0
6M	0.091	(0.069)	0.014	(0.02)	163	0.008
7M	0.078	(0.063)	-0.029	(0.022)	163	0.005
8M	0.051	(0.053)	-0.065***	(0.023)	163	0.002
9M	-0.051	(0.064)	-0.041*	(0.024)	163	0.002
10M	0.044	(0.068)	-0.018	(0.033)	163	0.001
11M	-0.037	(0.036)	0.024	(0.024)	163	0.001
12M	0.045	(0.114)	-0.005	(0.033)	161	0.001

TABLE 9: Brazil: Reaction of IP Forecasts, Means



Notes: The graphs show the annualized response of inflation expectations given a change in the unexpected component of the MPR decision and their confidence intervals. Each column corresponds to the days waited to measure the expectations after the meeting. Each graph shows the response of the current month (SM) until 17 months ahead (17M). The graphs in b, and c, show the coefficients of graph a, in blue.

FIGURE 11: Brazil: Change in Inflation Forecasts, 1-3 Days



Notes: The graphs show the response of GDP growth expectations given a change in the unexpected component of the MPR decision and their confidence intervals. Each column corresponds to the days waited to measure the expectations after the meeting. Each graph shows the previous quarter's (PQ) response until three quarters ahead (3Q). The graphs in b. and c, show the coefficients of graph a, in blue.





Notes: The graphs show the response of IP growth expectations given a change in the unexpected component of the MPR decision and their confidence intervals. Each column corresponds to the days waited to measure the expectations after the meeting. Each graph shows the response of the previous month (PM) until 12 months ahead (12M). The graphs in b. and c, show the coefficients of graph a, in blue.

FIGURE 13: Brazil: Change in IP Forecasts, 1-3 Days

A.2.2 Chile

This appendix shows the tables with all the regressions for Chile. The graphs in the main part of the paper show the coefficients of the regressions that include all controls, but this appendix shows the changes when adding one control each time.

The specification of the regressions using individual data is:

$$z^{e}_{d+s,t+j,i} - z^{e}_{d-m,t+j,i} = \alpha_{i,j} + \alpha_{j} + \sum_{k=0}^{K} \beta_{j,k} D_{k} (i_{t} - i^{e}_{d-m,t,i}) + C\left(_{\{d+s\}-\{d-m\}}\right) + \epsilon_{t,j,i}$$
(11)

The variables are the same as described in the main text but in their individual versions. As seen in equation (11), these results include individual fixed effects.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Surp. MPR	0.194^{**} (0.078)	0.200^{***} (0.070)	$\begin{array}{c} 0.257^{***} \\ (0.067) \end{array}$	$\begin{array}{c} 0.257^{***} \\ (0.067) \end{array}$	$\begin{array}{c} 0.253^{***} \\ (0.066) \end{array}$	$\begin{array}{c} 0.233^{***} \\ (0.071) \end{array}$	$\begin{array}{c} 0.234^{***} \\ (0.070) \end{array}$	$\begin{array}{c} 0.232^{***} \\ (0.071) \end{array}$
Surp. Infl.		$0.115 \\ (0.086)$	$0.100 \\ (0.093)$	$0.102 \\ (0.095)$	$0.108 \\ (0.093)$	$0.102 \\ (0.093)$	$0.099 \\ (0.093)$	$0.096 \\ (0.096)$
Δ IPSA			-0.389^{**} (0.186)	-0.407^{**} (0.190)	-0.310 (0.216)	-0.200 (0.219)	-0.210 (0.224)	-0.214 (0.229)
Δ NER				-0.081 (0.383)	-0.301 (0.427)	-0.176 (0.414)	-0.153 (0.418)	-0.159 (0.420)
Δ Copp.P.					-0.293 (0.270)	-0.270 (0.265)	-0.272 (0.268)	-0.267 (0.276)
Δ WTI						$0.018 \\ (0.082)$	$0.020 \\ (0.086)$	0.021 (0.086)
Δ IPE							$0.004 \\ (0.008)$	$0.004 \\ (0.008)$
Δ Imacec								$0.000 \\ (0.001)$
Constant	0.015^{*} (0.009)	0.011 (0.009)	$0.009 \\ (0.009)$	$0.009 \\ (0.009)$	$0.009 \\ (0.009)$	0.011 (0.009)	0.011 (0.009)	$0.011 \\ (0.009)$
N Adj. R^2	99 0.086	99 0.100	$99 \\ 0.131$	$99 \\ 0.123$	$99 \\ 0.127$	$\begin{array}{c} 98 \\ 0.093 \end{array}$	98 0.086	$98 \\ 0.075$

TABLE 10: Chile: Reaction to Next Month's Inflation Forecast, Median

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Surp. MPR	$\begin{array}{c} 0.187^{***} \\ (0.053) \end{array}$	$\begin{array}{c} 0.201^{***} \\ (0.030) \end{array}$	$\begin{array}{c} 0.218^{***} \\ (0.033) \end{array}$	$\begin{array}{c} 0.218^{***} \\ (0.034) \end{array}$	$\begin{array}{c} 0.215^{***} \\ (0.033) \end{array}$	$\begin{array}{c} 0.196^{***} \\ (0.034) \end{array}$	$\begin{array}{c} 0.197^{***} \\ (0.034) \end{array}$	$\begin{array}{c} 0.158^{***} \\ (0.037) \end{array}$
Surp. Infl		$\begin{array}{c} 0.185^{***} \\ (0.044) \end{array}$	$\begin{array}{c} 0.183^{***} \\ (0.043) \end{array}$	0.185^{***} (0.046)	$\begin{array}{c} 0.187^{***} \\ (0.045) \end{array}$	0.177^{***} (0.048)	$\begin{array}{c} 0.179^{***} \\ (0.049) \end{array}$	$\begin{array}{c} 0.144^{***} \\ (0.047) \end{array}$
Δ IPSA			-0.106 (0.152)	-0.113 (0.161)	$0.041 \\ (0.194)$	$0.124 \\ (0.196)$	$0.129 \\ (0.201)$	$0.073 \\ (0.212)$
Δ NER				-0.043 (0.400)	-0.221 (0.373)	-0.122 (0.385)	-0.136 (0.383)	-0.185 (0.355)
Δ Copp.P.					-0.408 (0.324)	-0.400 (0.354)	-0.379 (0.372)	-0.208 (0.384)
Δ WTI						$0.021 \\ (0.047)$	$0.015 \\ (0.056)$	$\begin{array}{c} 0.016 \\ (0.054) \end{array}$
Δ IPE							-0.003 (0.007)	-0.002 (0.008)
Δ Imacec								0.003^{**} (0.001)
Constant	-0.003 (0.012)	-0.019 (0.011)	-0.021^{*} (0.012)	-0.021^{*} (0.012)	-0.020 (0.012)	-0.015 (0.013)	-0.015 (0.013)	-0.019 (0.013)
N Adj. R^2	$37 \\ 0.255$	$\begin{array}{c} 37 \\ 0.439 \end{array}$	$\begin{array}{c} 37 \\ 0.430 \end{array}$	$\begin{array}{c} 37\\ 0.413\end{array}$	$\begin{array}{c} 37 \\ 0.430 \end{array}$	$\begin{array}{c} 36 \\ 0.407 \end{array}$	$\frac{36}{0.387}$	$\frac{36}{0.444}$

TABLE 11: Chile: Reaction of One-Month's ahead Inflation Forecast, Median

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Surp. MPR	0.398^{*} (0.227)	0.398^{*} (0.226)	0.464^{*} (0.249)	0.461^{*} (0.245)	0.456^{*} (0.243)	$0.390 \\ (0.260)$	$0.387 \\ (0.261)$	$0.348 \\ (0.243)$
Surp. Infl.		-0.001 (0.125)	-0.012 (0.124)	-0.008 (0.122)	-0.003 (0.120)	-0.013 (0.120)	-0.013 (0.120)	-0.029 (0.120)
Δ IPSA			-0.502 (0.469)	-0.681 (0.553)	-0.594 (0.529)	-0.290 (0.423)	-0.303 (0.427)	-0.346 (0.434)
Δ NER				-0.835 (1.087)	-1.009 (1.311)	-0.640 (1.289)	-0.616 (1.297)	-0.711 (1.312)
Δ Copp.P.					-0.252 (0.532)	-0.269 (0.430)	-0.274 (0.433)	-0.203 (0.422)
Δ WTI						$0.141 \\ (0.222)$	$0.147 \\ (0.221)$	$0.146 \\ (0.220)$
Δ IPE							$0.005 \\ (0.016)$	$0.004 \\ (0.016)$
Δ Imacec								$0.003 \\ (0.004)$
Constant	-0.004 (0.017)	-0.004 (0.018)	-0.006 (0.019)	-0.003 (0.019)	-0.003 (0.020)	$0.002 \\ (0.019)$	$0.002 \\ (0.019)$	-0.008 (0.019)
N Adj. R^2	$\begin{array}{c} 122 \\ 0.068 \end{array}$	$\begin{array}{c} 122 \\ 0.060 \end{array}$	$\begin{array}{c} 122 \\ 0.065 \end{array}$	$\begin{array}{c} 122 \\ 0.066 \end{array}$	$\begin{array}{c} 122 \\ 0.060 \end{array}$	$\begin{array}{c} 121 \\ 0.042 \end{array}$	$\begin{array}{c} 121 \\ 0.034 \end{array}$	$\begin{array}{c} 121 \\ 0.033 \end{array}$

 TABLE 12: Chile: Reaction of One-year's Ahead Inflation Forecast, Median

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Surp. MPR	0.034 (0.141)	0.051 (0.115)	0.063 (0.133)	0.063 (0.133)	0.065 (0.133)	0.074 (0.138)	0.067 (0.135)	0.024 (0.137)
Surp. Infl.		$\begin{array}{c} 0.178^{*} \\ (0.091) \end{array}$	$\begin{array}{c} 0.176^{*} \ (0.093) \end{array}$	$\begin{array}{c} 0.176^{*} \ (0.093) \end{array}$	$\begin{array}{c} 0.173^{*} \\ (0.093) \end{array}$	$\begin{array}{c} 0.176^{*} \ (0.093) \end{array}$	0.175^{*} (0.089)	0.158^{**} (0.079)
Δ IPSA			-0.090 (0.190)	-0.106 (0.237)	-0.157 (0.262)	$0.014 \\ (0.290)$	-0.022 (0.265)	-0.069 (0.258)
Δ NER				-0.075 (0.484)	$0.026 \\ (0.586)$	$\begin{array}{c} 0.117 \\ (0.563) \end{array}$	$0.184 \\ (0.561)$	$\begin{array}{c} 0.080 \\ (0.550) \end{array}$
Δ Copp.P.					$0.146 \\ (0.257)$	$0.289 \\ (0.230)$	$0.274 \\ (0.233)$	$\begin{array}{c} 0.353 \ (0.238) \end{array}$
Δ WTI						-0.148 (0.120)	-0.131 (0.109)	-0.132 (0.104)
Δ IPE							0.015^{*} (0.009)	0.014 (0.009)
Δ Imacec								0.004^{*} (0.002)
Constant	$0.002 \\ (0.009)$	-0.006 (0.008)	-0.006 (0.008)	-0.006 (0.009)	-0.006 (0.009)	-0.005 (0.009)	-0.005 (0.009)	-0.016 (0.010)
N Adj. R^2	122 -0.006	$\begin{array}{c} 122\\ 0.084 \end{array}$	$\begin{array}{c} 122 \\ 0.078 \end{array}$	$\begin{array}{c} 122 \\ 0.070 \end{array}$	$\begin{array}{c} 122 \\ 0.065 \end{array}$	$\begin{array}{c} 121 \\ 0.074 \end{array}$	$\begin{array}{c} 121 \\ 0.094 \end{array}$	$\begin{array}{c} 121 \\ 0.122 \end{array}$

TABLE 13: Chile: Reaction of Two-Years' ahead Inflation Forecast, Median

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Surp. MPR	0.263^{**} (0.101)	$\begin{array}{c} 0.269^{***} \\ (0.086) \end{array}$	$\begin{array}{c} 0.324^{***} \\ (0.065) \end{array}$	$\begin{array}{c} 0.323^{***} \\ (0.065) \end{array}$	$\begin{array}{c} 0.317^{***} \\ (0.060) \end{array}$	$\begin{array}{c} 0.297^{***} \\ (0.068) \end{array}$	$\begin{array}{c} 0.295^{***} \\ (0.068) \end{array}$	$\begin{array}{c} 0.301^{***} \\ (0.070) \end{array}$
Surp. Infl.		$\begin{array}{c} 0.135^{*} \\ (0.080) \end{array}$	$0.117 \\ (0.087)$	$0.121 \\ (0.090)$	$0.130 \\ (0.087)$	$0.126 \\ (0.086)$	$0.123 \\ (0.085)$	$0.128 \\ (0.088)$
Δ IPSA			-0.387^{**} (0.189)	-0.425^{**} (0.176)	-0.296 (0.205)	-0.220 (0.239)	-0.228 (0.242)	-0.221 (0.248)
Δ NER				-0.172 (0.417)	-0.464 (0.415)	-0.392 (0.417)	-0.368 (0.422)	-0.358 (0.424)
Δ Copp.P.					-0.390 (0.273)	-0.373 (0.278)	-0.375 (0.280)	-0.383 (0.285)
Δ WTI						$0.004 \\ (0.092)$	$0.007 \\ (0.096)$	$0.005 \\ (0.096)$
Δ IPE							$0.004 \\ (0.008)$	$0.004 \\ (0.008)$
Δ Imacec								-0.000 (0.001)
Constant	$0.012 \\ (0.009)$	$0.007 \\ (0.008)$	$0.005 \\ (0.008)$	$0.006 \\ (0.007)$	$0.006 \\ (0.007)$	$0.007 \\ (0.008)$	$0.007 \\ (0.008)$	$0.008 \\ (0.008)$
$\frac{N}{\text{Adj. }R^2}$	99 0.154	99 0.183	99 0.219	99 0.213	99 0.230	98 0.178	98 0.172	98 0.163

TABLE 14: Chile: Reaction of Next Month's Ahead Inflation Forecast, Mean

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Surp. MPR	$\begin{array}{c} 0.165^{***} \\ (0.050) \end{array}$	$\begin{array}{c} 0.169^{***} \\ (0.035) \end{array}$	$\begin{array}{c} 0.176^{***} \\ (0.044) \end{array}$	$\begin{array}{c} 0.176^{***} \\ (0.044) \end{array}$	$\begin{array}{c} 0.175^{***} \\ (0.043) \end{array}$	$\begin{array}{c} 0.144^{***} \\ (0.043) \end{array}$	$\begin{array}{c} 0.147^{***} \\ (0.043) \end{array}$	0.103^{*} (0.051)
Surp. Infl.		$\begin{array}{c} 0.126^{***} \\ (0.031) \end{array}$	$\begin{array}{c} 0.125^{***} \\ (0.029) \end{array}$	$\begin{array}{c} 0.124^{***} \\ (0.034) \end{array}$	$\begin{array}{c} 0.128^{***} \\ (0.033) \end{array}$	$\begin{array}{c} 0.116^{***} \\ (0.039) \end{array}$	$\begin{array}{c} 0.118^{***} \\ (0.040) \end{array}$	$\begin{array}{c} 0.085^{**} \\ (0.038) \end{array}$
Δ IPSA			-0.039 (0.168)	-0.034 (0.189)	$0.106 \\ (0.193)$	$0.261 \\ (0.184)$	$0.264 \\ (0.185)$	$0.206 \\ (0.183)$
Δ NER				0.027 (0.286)	-0.139 (0.265)	$0.005 \\ (0.286)$	-0.009 (0.291)	-0.050 (0.239)
Δ Copp.P.					-0.376 (0.245)	-0.282 (0.259)	-0.260 (0.297)	-0.086 (0.244)
Δ WTI						-0.024 (0.047)	-0.031 (0.065)	-0.027 (0.060)
Δ IPE							-0.003 (0.010)	-0.002 (0.009)
Δ Imacec								0.003^{**} (0.001)
Constant	$0.011 \\ (0.010)$	-0.000 (0.010)	-0.001 (0.011)	-0.001 (0.012)	-0.000 (0.011)	$0.008 \\ (0.012)$	0.008 (0.012)	0.003 (0.011)
$\frac{N}{\text{Adj. }R^2}$	$37 \\ 0.252$	37 0.380	37 0.363	37 0.343	$37 \\ 0.367$	$\frac{36}{0.334}$	$\frac{36}{0.313}$	36 0.409

TABLE 15: Chile: Reaction of One-Month's ahead Inflation Forecast, Mean

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Surp. MPR	0.389^{*} (0.213)	0.392^{*} (0.210)	0.455^{**} (0.224)	$\begin{array}{c} 0.453^{**} \\ (0.222) \end{array}$	$\begin{array}{c} 0.450^{**} \\ (0.221) \end{array}$	$0.346 \\ (0.242)$	$0.346 \\ (0.238)$	0.284 (0.218)
Surp. Infl.		$0.063 \\ (0.100)$	$\begin{array}{c} 0.047 \\ (0.100) \end{array}$	$0.048 \\ (0.100)$	$\begin{array}{c} 0.052 \\ (0.099) \end{array}$	$\begin{array}{c} 0.046 \\ (0.099) \end{array}$	$\begin{array}{c} 0.046 \\ (0.099) \end{array}$	$0.026 \\ (0.097)$
Δ IPSA			-0.501 (0.401)	-0.543 (0.454)	-0.470 (0.457)	-0.240 (0.411)	-0.239 (0.414)	-0.300 (0.420)
Δ NER				-0.194 (0.971)	-0.338 (1.131)	-0.081 (1.133)	-0.083 (1.140)	-0.214 (1.143)
Δ Copp.P.					-0.209 (0.477)	-0.279 (0.406)	-0.279 (0.407)	-0.181 (0.384)
Δ WTI						$0.166 \\ (0.197)$	$0.165 \\ (0.196)$	$0.168 \\ (0.197)$
Δ IPE							-0.000 (0.014)	-0.001 (0.014)
Δ Imacec								$0.005 \\ (0.004)$
Constant	-0.002 (0.014)	-0.005 (0.016)	-0.007 (0.016)	-0.006 (0.016)	-0.006 (0.016)	-0.001 (0.016)	-0.001 (0.016)	-0.014 (0.017)
N Adj. R^2	$122 \\ 0.077$	$122 \\ 0.073$	$\begin{array}{c} 122\\ 0.082 \end{array}$	$122 \\ 0.075$	$122 \\ 0.069$	$\begin{array}{c} 121 \\ 0.036 \end{array}$	$121 \\ 0.027$	$\begin{array}{c} 121 \\ 0.037 \end{array}$

TABLE 16: Chile: Reaction of One-Year's Ahead Inflation Forecast, Mean

$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$
$\begin{array}{cccccccccccccccccccccccccccccccccccc$
Surp. Infl. 0.142^{**} (0.064) 0.143^{**} (0.066) 0.146^{**} (0.066) 0.145^{**} (0.066) 0.145^{**} (0.063) 0.131^{**} (0.056) Δ IPSA 0.042 (0.145) -0.015 (0.164) 0.071 (0.177) 0.057 (0.209) 0.016 (0.202) Δ NER -0.267 (0.321) -0.273 (0.392) -0.239 (0.398) -0.239 (0.400) -0.326 (0.394) Δ Copp.P. -0.267 (0.321) -0.075 (0.196) -0.063 (0.183) 0.003 (0.187) Δ WTI -0.075 (0.088) -0.003 (0.183) -0.003 (0.079) -0.003 (0.079) Δ IPE -0.018 (0.007) -0.008 (0.007) -0.008 (0.007)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$
$\begin{array}{cccccccccccccccccccccccccccccccccccc$
$ \begin{array}{c cccc} \Delta \ \mathrm{NER} & -0.267 \\ (0.321) & 0.319 \\ (0.392) & 0.398 \\ (0.398) & 0.400 \\ (0.400) & 0.394 \\ (0.394) \\ \end{array} \\ \begin{array}{c ccccccccccccccccccccccccccccccccccc$
$\begin{array}{cccccccccccccccccccccccccccccccccccc$
$\begin{array}{cccc} \Delta \ {\rm Copp.P.} & -0.075 \\ \Delta \ {\rm WTI} & & 0.003 \\ \Delta \ {\rm WTI} & & -0.013 \\ \Delta \ {\rm IPE} & & 0.008 \\ \end{array} \begin{array}{c} -0.003 \\ -0.003 \\ (0.088) \\ 0.088 \\ 0.007 \\ (0.007) \end{array} \begin{array}{c} -0.003 \\ (0.088) \\ 0.007 \\ (0.007) \\ (0.007) \end{array}$
$\begin{array}{ccccccc} (0.196) & (0.183) & (0.187) & (0.194) \\ \Delta \ WTI & & -0.013 & -0.003 & -0.001 \\ & (0.088) & (0.083) & (0.079) \\ \Delta \ IPE & & & & & & & & \\ & & & & & & & & & \\ & & & & & & & & & \\ & & & & & & & & & & \\ & & & & & & & & & & \\ & & & & & & & & & & \\ & & & & & & & & & & \\ & & & & & & & & & & \\ & & & & & & & & & & \\ & & & & & & & & & \\ & & & & & & & & & \\ & & & & & & & & & \\ & & & & & & & & & \\ & & & & & & & & \\ & & & & & & & & \\ & & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & \\ & & & & & \\ & & & & & $
$ \begin{array}{c} \Delta \text{ WTI} & -0.013 & -0.003 & -0.001 \\ (0.088) & (0.083) & (0.079) \\ \end{array} \\ \Delta \text{ IPE} & & 0.008 & 0.007 \\ (0.007) & (0.007) \end{array} $
$\begin{array}{cccc} (0.088) & (0.083) & (0.079) \\ & \Delta \text{ IPE} & & 0.008 & 0.007 \\ & & (0.007) & (0.007) \end{array}$
$\begin{array}{ccc} \Delta \text{ IPE} & & 0.008 & 0.007 \\ (0.007) & (0.007) \end{array}$
(0.007) (0.007)
Δ Imacec 0.003^*
(0.002)
Constant 0.005 -0.001 -0.001 -0.000 -0.000 0.001 0.000 -0.008
(0.007) (0.007) (0.007) (0.007) (0.007) (0.007) (0.007) (0.007) (0.008)
N 122 122 122 122 122 121 121 121
Adj. R^2 0.018 0.103 0.096 0.094 0.087 0.074 0.077 0.107

TABLE 17: Chile: Reaction of Two-Years' ahead Inflation Forecast, Mean

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Surp. MPR	$\begin{array}{c} 0.119^{***} \\ (0.032) \end{array}$	$\begin{array}{c} 0.119^{***} \\ (0.032) \end{array}$	$\begin{array}{c} 0.128^{***} \\ (0.038) \end{array}$	$\begin{array}{c} 0.128^{***} \\ (0.037) \end{array}$	$\begin{array}{c} 0.126^{***} \\ (0.037) \end{array}$	$\begin{array}{c} 0.109^{***} \\ (0.033) \end{array}$	$\begin{array}{c} 0.109^{***} \\ (0.033) \end{array}$	$\begin{array}{c} 0.105^{***} \\ (0.033) \end{array}$
Surp. Infl.		$\begin{array}{c} 0.113^{***} \\ (0.020) \end{array}$	$\begin{array}{c} 0.106^{***} \\ (0.020) \end{array}$	$\begin{array}{c} 0.109^{***} \\ (0.020) \end{array}$	$\begin{array}{c} 0.120^{***} \\ (0.020) \end{array}$	$\begin{array}{c} 0.110^{***} \\ (0.020) \end{array}$	0.106^{***} (0.020)	$\begin{array}{c} 0.099^{***} \\ (0.021) \end{array}$
Δ IPSA			-0.185^{***} (0.053)	-0.212^{***} (0.048)	-0.052 (0.072)	$0.020 \\ (0.069)$	$0.011 \\ (0.067)$	-0.006 (0.066)
Δ NER				-0.118 (0.116)	-0.487^{***} (0.075)	-0.351^{***} (0.070)	-0.324^{***} (0.069)	-0.340^{***} (0.072)
Δ Copp.P.					-0.508^{***} (0.136)	-0.519^{***} (0.142)	-0.524^{***} (0.143)	-0.505^{***} (0.145)
Δ WTI						0.068^{***} (0.021)	$\begin{array}{c} 0.072^{***} \\ (0.022) \end{array}$	0.072^{***} (0.022)
Δ IPE							0.005^{**} (0.002)	0.005^{**} (0.002)
Δ Imacec								0.001^{*} (0.000)
Constant	0.017^{***} (0.000)	$\begin{array}{c} 0.013^{***} \\ (0.001) \end{array}$	$\begin{array}{c} 0.012^{***} \\ (0.001) \end{array}$	$\begin{array}{c} 0.013^{***} \\ (0.001) \end{array}$	$\begin{array}{c} 0.012^{***} \\ (0.001) \end{array}$	$\begin{array}{c} 0.014^{***} \\ (0.001) \end{array}$	$\begin{array}{c} 0.014^{***} \\ (0.001) \end{array}$	$\begin{array}{c} 0.012^{***} \\ (0.001) \end{array}$
N Adj. R^2	$4,718 \\ 0.029$	$4,718 \\ 0.039$	$4,718 \\ 0.042$	$4,718 \\ 0.042$	$4,718 \\ 0.055$	$4,671 \\ 0.047$	$4,671 \\ 0.048$	$4,671 \\ 0.048$

TABLE 18: Chile: Reaction of Next Month's Inflation Forecasts, Individual Data

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Surp. MPR	$\begin{array}{c} 0.099^{***} \\ (0.020) \end{array}$	$\begin{array}{c} 0.098^{***} \\ (0.020) \end{array}$	$\begin{array}{c} 0.095^{***} \\ (0.020) \end{array}$	$\begin{array}{c} 0.095^{***} \\ (0.020) \end{array}$	$\begin{array}{c} 0.093^{***} \\ (0.020) \end{array}$	0.078^{***} (0.015)	$\begin{array}{c} 0.078^{***} \\ (0.015) \end{array}$	$\begin{array}{c} 0.063^{***} \\ (0.009) \end{array}$
Surp. Infl.		$\begin{array}{c} 0.109^{***} \\ (0.019) \end{array}$	$\begin{array}{c} 0.110^{***} \\ (0.020) \end{array}$	$\begin{array}{c} 0.111^{***} \\ (0.020) \end{array}$	$\begin{array}{c} 0.116^{***} \\ (0.020) \end{array}$	$\begin{array}{c} 0.103^{***} \\ (0.020) \end{array}$	$\begin{array}{c} 0.102^{***} \\ (0.020) \end{array}$	$\begin{array}{c} 0.074^{***} \\ (0.020) \end{array}$
Δ IPSA			$0.057 \\ (0.047)$	$0.053 \\ (0.048)$	$\begin{array}{c} 0.181^{***} \\ (0.062) \end{array}$	$\begin{array}{c} 0.320^{***} \\ (0.068) \end{array}$	$\begin{array}{c} 0.317^{***} \\ (0.068) \end{array}$	$\begin{array}{c} 0.225^{***} \\ (0.069) \end{array}$
Δ NER				-0.020 (0.105)	-0.175 (0.108)	$0.003 \\ (0.106)$	$0.011 \\ (0.104)$	-0.032 (0.104)
Δ Copp.P.					-0.348^{***} (0.086)	-0.291^{***} (0.087)	-0.302^{***} (0.088)	-0.109 (0.092)
Δ WTI						$0.009 \\ (0.018)$	$0.012 \\ (0.018)$	$0.005 \\ (0.018)$
Δ IPE							$0.002 \\ (0.002)$	$0.002 \\ (0.002)$
Δ Imacec								0.003^{***} (0.000)
Constant	0.015^{***} (0.000)	0.006^{***} (0.002)	0.007^{***} (0.002)	0.007^{***} (0.002)	$\begin{array}{c} 0.007^{***} \\ (0.002) \end{array}$	$\begin{array}{c} 0.014^{***} \\ (0.002) \end{array}$	$\begin{array}{c} 0.014^{***} \\ (0.002) \end{array}$	0.007^{***} (0.002)
N Adj. R^2	$1,835 \\ 0.060$	$1,835 \\ 0.091$	$1,835 \\ 0.091$	$1,835 \\ 0.091$	$1,835 \\ 0.098$	$1,788 \\ 0.093$	$1,788 \\ 0.093$	$1,788 \\ 0.118$

TABLE 19: Chile: Reaction of One-Month's ahead Inflation Forecasts, Individual Data

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Surp. MPR	$\begin{array}{c} 0.189^{***} \\ (0.066) \end{array}$	$\begin{array}{c} 0.188^{***} \\ (0.066) \end{array}$	$\begin{array}{c} 0.200^{***} \\ (0.073) \end{array}$	$\begin{array}{c} 0.200^{***} \\ (0.073) \end{array}$	$\begin{array}{c} 0.199^{***} \\ (0.073) \end{array}$	$\begin{array}{c} 0.146^{**} \\ (0.062) \end{array}$	$\begin{array}{c} 0.144^{**} \\ (0.061) \end{array}$	0.117^{**} (0.054)
Surp. Infl.		$\begin{array}{c} 0.091^{***} \\ (0.033) \end{array}$	0.085^{**} (0.034)	0.085^{**} (0.034)	0.090^{**} (0.034)	0.083^{**} (0.033)	0.083^{**} (0.033)	0.063^{*} (0.033)
Δ IPSA			-0.240^{*} (0.125)	-0.223^{*} (0.130)	-0.110 (0.142)	$0.082 \\ (0.142)$	$0.066 \\ (0.141)$	-0.043 (0.135)
Δ NER				$0.076 \\ (0.303)$	-0.150 (0.306)	$0.192 \\ (0.298)$	$0.218 \\ (0.293)$	$0.080 \\ (0.299)$
Δ Copp.P.					-0.340^{**} (0.132)	-0.431^{***} (0.139)	-0.444^{***} (0.139)	-0.305^{**} (0.135)
Δ WTI						$\begin{array}{c} 0.237^{***} \\ (0.073) \end{array}$	$\begin{array}{c} 0.246^{***} \\ (0.072) \end{array}$	0.236^{***} (0.072)
Δ IPE							$0.007 \\ (0.005)$	$0.005 \\ (0.004)$
Δ Imacec								0.006^{***} (0.001)
Constant	0.009^{***} (0.001)	0.005^{**} (0.002)	0.004^{**} (0.002)	0.004^{*} (0.002)	0.005^{**} (0.002)	$\begin{array}{c} 0.009^{***} \\ (0.002) \end{array}$	$\begin{array}{c} 0.009^{***} \\ (0.002) \end{array}$	-0.008** (0.003)
N Adj. R^2	$5,790 \\ 0.014$	$5,790 \\ 0.017$	$5,790 \\ 0.017$	$5,790 \\ 0.017$	$5,790 \\ 0.018$	$5,743 \\ 0.019$	$5,743 \\ 0.019$	$5,743 \\ 0.025$

TABLE 20: Chile: Reaction of One-Year's ahead Inflation Forecasts, Individual Data

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Surp. MPR	0.066^{**} (0.027)	0.066^{**} (0.027)	0.062^{**} (0.027)	0.062^{**} (0.027)	0.062^{**} (0.027)	0.059^{**} (0.027)	0.056^{**} (0.026)	0.042^{*} (0.023)
Surp. Infl.		$\begin{array}{c} 0.103^{***} \\ (0.028) \end{array}$	$\begin{array}{c} 0.105^{***} \\ (0.028) \end{array}$	$\begin{array}{c} 0.106^{***} \\ (0.028) \end{array}$	$\begin{array}{c} 0.106^{***} \\ (0.029) \end{array}$	$\begin{array}{c} 0.106^{***} \\ (0.029) \end{array}$	$\begin{array}{c} 0.105^{***} \\ (0.029) \end{array}$	$\begin{array}{c} 0.095^{***} \\ (0.028) \end{array}$
Δ IPSA			$0.075 \\ (0.092)$	$0.064 \\ (0.093)$	$0.079 \\ (0.106)$	$0.169 \\ (0.115)$	$0.149 \\ (0.113)$	$0.092 \\ (0.115)$
Δ NER				-0.048 (0.200)	-0.078 (0.227)	-0.005 (0.219)	0.027 (0.216)	-0.046 (0.221)
Δ Copp.P.					-0.045 (0.115)	-0.001 (0.100)	-0.017 (0.101)	$0.056 \\ (0.101)$
Δ WTI						-0.033 (0.049)	-0.023 (0.048)	-0.028 (0.048)
Δ IPE							0.008^{**} (0.004)	0.008^{**} (0.004)
Δ Imacec								0.003^{***} (0.001)
Constant	0.008^{***} (0.000)	0.004^{**} (0.001)	0.004^{***} (0.001)	0.004^{**} (0.002)	0.004^{**} (0.002)	0.005^{***} (0.002)	0.005^{***} (0.002)	-0.004 (0.003)
N Adj. R^2	$5,790 \\ 0.008$	$5,790 \\ 0.014$	$5,790 \\ 0.014$	$5,790 \\ 0.014$	$5,790 \\ 0.014$	$5,743 \\ 0.014$	$5,743 \\ 0.015$	$5,743 \\ 0.018$

TABLE 21: Chile: Reaction of Two-Years' ahead Inflation Forecasts, Individual Data

A.3 Appendix to the Additional Results of the Empirical Section

The exercises of this section use dummies to allow for different responses when the surprise in the interest rate occurs under different circumstances. The general regressions for the first part are of the form:

$$z_{d+s,t+j}^e - z_{d-m,t+j}^e = \alpha_j + \sum_{k=0}^K \beta_{j,k} D_k (i_t - i_{d-m,t}^e) + C\left(_{\{d+s\}-\{d-m\}}\right) + \epsilon_{t,j}$$
(12)

where the variables are the same as in the main text and D_k are dummies that vary

according to the exercise the following way:

- Dependence on the Movement of the MPR: D_0 is equal to one when there is no movement in the MPR, zero otherwise; D_1 is equal to one when the movement in the MPR goes in the opposite direction as the surprise, zero otherwise; D_2 is equal to one when the movement in the MPR goes in the same direction as the surprise, zero otherwise.
- Dependence on the response of the stock market: D_0 is equal to one when the daily movement in the stock market index the day after the meeting goes in the opposite direction as the surprise, zero otherwise; D_1 is equal to one when the daily movement in the stock market index the day after the meeting goes in the same direction as the surprise, zero otherwise.
- Dependence on the Intensity of the surprise: D_0 is equal to one when the surprise is small (below 50 basis points) and zero otherwise; D_1 is equal to one when the MPR surprise is 50 basis points or above and zero otherwise.
- Subsamples: There are different exercises in this part. On the one hand, some excluded part of the sample to get results (the periods at the effective lower bound or the unstable periods in Brazil), while others defined different dummies.
 - In the exercise for Chile to differentiate when there is an IPoM and not, D_0 is equal to one when there is an IPoM at the same time of the MPR decision and zero otherwise; D_1 is equal to one when there is no IPoM and zero otherwise.
 - In the exercise for Chile to differentiate before and after 2018, D_0 is equal to one when before 2018 and zero otherwise; D_1 is equal when the year is 2018 or above and zero otherwise.



Notes: The graphs show the response of Inflation, GDP, and IP expectations, given a change in the unexpected component of the MPR and its confidence intervals. The graphs separate the reaction when the surprise is small (column a.) and big (higher or equal to 50 basis points, column b.). For Inflation, the graphs show the response from the same month of the decision (SM) until 17 months ahead (17M), for GDP from the previous quarter (PQ) until three quarters ahead (3Q), and for IP from the previous month (PM) until 12 months ahead (12M). The graphs show in blue the baseline results from Figures 3 and 5.

FIGURE 14: Brazil: Dependence on the Intensity of the Surprise

A.3.1 Intensity of Surprise



Notes: The graphs show the response of Inflation expectations given a change in the unexpected component of the MPR and its confidence intervals. The graphs separate the reaction when the surprise is small (column a.) and big (higher or equal to 50 basis points, column b.). The graphs show the response for the same month, one month ahead, the first year ahead, and the second year ahead. The graphs show the baseline results from Figure 4 in red. All the regressions include all the controls, and the ones with individual data have clusterized standard errors and individual fixed effects. The coefficients of interest are the only ones included.

FIGURE 15: Chile: Dependence on the Intensity of the Surprise

A.3.2 Different subsamples



Notes: The graphs show the response of Inflation, GDP, and IP expectations, given a change in the unexpected component of the MPR and its confidence intervals. The graphs show the results for subsamples; the first excludes the years before 2004, and the second excludes unstable periods: the years before 2003 and between mid-2015 and September 2016. For Inflation, the graphs show the response from the same month of the decision (SM) until 17 months ahead (17M), for GDP from the previous quarter (PQ) until three quarters ahead (3Q), and for IP from the previous month (PM) until 12 months ahead (12M). The graphs show in blue the baseline results from Figures 3 and 5.





Notes: The graphs show the annualized response of inflation expectations given a change in the unexpected component of the MPR decision and their confidence intervals. Each graph shows the response of the following month, one month ahead, one year ahead, and two years ahead. The regressions include all the controls. Graph a. shows the response without excluding the ELB in red, and the ones in b. and c. show the coefficients of graph a. in red.





Notes: The graphs show the annualized response of inflation expectations given a change in the unexpected component of the MPR decision and their confidence intervals. Graphs in a. show the results for surprises before 2018, and the ones in b. for the surprise after 2018. The graphs on the left show the response of the following month, one year and two years ahead. The graphs on the right also include the 1-month ahead (only available starting in 2018). The regressions include all the controls.

FIGURE 18: Chile: Differentiating between Surprises before and after 2018



Notes: The graphs show the annualized response of inflation expectations given a change in the unexpected component of the MPR decision and their confidence intervals. The graphs in a. show when the surprise does not coincide with the release of an IPoM, while the ones in b. show cases when it does coincide. Each graph shows the response of the following month, one month ahead, one year ahead, and two years ahead. The regressions include all the controls.

FIGURE 19: Chile: Differentiating between Surprise with and without IPoM

A.3.3 Different measures



Notes: The graphs show the annualized response of inflation expectations given a change in the unexpected component of the MPR decision and their confidence intervals. The graphs on the left show the response using the complete sample (2010-2022), and the ones on the right exclude the period at the ELB (the red line in these graphs corresponds to the results of the whole sample). The graphs on the top use break-even inflation as the measure, and the ones on the left use Inflation Insurance. In all cases the regressions include all the controls.

FIGURE 20: Chile: Other Inflation Measures

Using break-even inflation based on nominal and real Swaps

This measure uses Swaps at different horizons. The ones reported in the graph are one year (1Y), 1-in-1 (2Y, 1 year but starting one year in the future), 1-in-2 (3Y, 1 year starting two years in the future), and 5-in-5 (LT, 5 years starting five years in the future).

The calculations are based on the instruments called "Swap Promedio Cámara" in Pesos and UF, and breakeven inflation is the difference between the two.²⁵ The inflation estimated from these documents corresponds to an average of inflation during the period under study. For more information, refer to Varela (2007).

Using Inflation Insurance

These are financial instruments in which one side pays a fixed inflation (expressed over

²⁵The UF means "Unidad de Fomento" which is an indexed measure of account in Chile.

the value of UF) over some capital and horizon, and the counterpart pays actual inflation. With these in hand, one gets expectations of accumulated inflation and expectations at different horizons. For more information, refer to .

A.4 Appendix to Motivating Model

A.4.1 Environment and Problems of Every Agent

Households

Assume there is a representative household that seeks to maximize the discounted sum of utilities of the form:

$$\mathbb{E}_t \left\{ \sum_{T=t}^{\infty} \beta^{T-t} \left[\frac{C_T^{1-1/\sigma}}{1-1/\sigma} - \frac{H_T^{1+1/\xi}}{1+1/\xi} \right] \right\}$$

 C_t is consumption, and H_t is the quantity of labor supplied in time t. In this formulation, σ is the elasticity of intertemporal substitution, and ξ is the Frisch elasticity of labor supply. The index C_t is a Dixit-Stiglitz aggregate of differentiated consumer goods:

$$C_t = \left[\int_0^1 C_t(j)^{(\theta_t - 1)/\theta_t} dj\right]^{\theta_t/(\theta_t - 1)}$$

with $\theta_t > 1$ the elasticity of substitution between varieties, which is a stochastic shock with mean $\mathbb{E}(\theta_t) = \overline{\theta}$. The shock to the elasticity allows for a cost-push shock. The price index that defines the minimum cost of a unit of C_t is:

$$P_t = \left[\int_0^1 P_t(j)^{1-\theta_t} dj\right]^{1/(1-\theta_t)}$$

)

For simplicity, assume only one traded asset each period: a one-period riskless nominal bond, which allows the central bank to set that interest rate. The households own the firms (same share each household) and face a tax set by the government. So, its budget constraint in t is:

$$P_t C_t + B_t = (1 + \tilde{i}_{t-1})B_{t-1} + W_t H_t + \int_0^1 \Pi_t(j)dj - T_t$$

where B_t is the one period nominal bond that pays \tilde{i}_t in period t + 1, W_t is the nominal wage determined in the labor market, $\Pi_t(i)$ are the nominal profits of firm j and T_t is a tax set by the government.

Each household is the same, and so they make the same decisions. Households choose

the amount consumed, worked, and saved, taking prices as given. So their three optimality conditions are: (i) optimality across differentiated goods, (ii) intertemporal optimality, which is the Euler equation, and (iii) optimal supply of labor.

Firms

Each firm j is the monopolistic supplier of good (j), so given the household demand, it chooses its price. Its production function uses only labor as input:

$$Y_t(j) = A_t H_t(j)$$

Where A_t is a productivity factor common to all firms. The profits of each firm j are:

$$\Pi_t(P_t(j), .) = P_t(j)Y_t(j) - W_tH_t(j)$$

Additionally, assume as Calvo (1983) that they can only choose their price in any given period with probability $(1 - \alpha)$. Firms that choose their price maximize the present discounted value of profits in all future states prior to the next reconsideration of its price:

$$\hat{\mathbb{E}}_t \left\{ \sum_{s=0}^{\infty} \alpha^s Q_{t,t+s} \Pi_{t+s}(P_t(j)) \right\}$$

where $Q_{t,t+s} = \beta^s [U'(C_{t+s})/U'(C_t)] [P_t/P_{t+s}]$ is the stochastic discount factor of the household.

Government

The government is assumed to consume the same composite of the goods demanded by households:

$$G_t = \left[\int_0^1 G_t(i)^{(\theta_t - 1)/\theta_t} di\right]^{\theta_t/(\theta_t - 1)}$$

where $G_t(i)$ is the quantity consumed of good *i* and G_t is the Dixit-Stiglitz aggregate. It is also assumed to hold a balanced budget:

$$T_t = P_t G_t$$

$Central \ Bank$

The Central Bank sets the interest rate following a Taylor rule of the form:

$$(1+\tilde{i}_t) = \bar{i} \left(\frac{P_t}{P_{t-1}}\right)^{\phi_{\pi}} \left(\frac{Y_t}{Y_t^*}\right)^{\phi_x} e^{\epsilon_t}$$

Where Y_t^* is the efficient level of output in period t, ϵ_t is the monetary shock in period tand ϕ_{π} , ϕ_x are such that the Taylor principle is satisfied.

Market Clearing

In Equilibrium the goods market, the bonds market and the labor market clear, which means:

$$Y_t = C_t + G_t$$
$$H_t = \int_0^1 H_t(i) di$$
$$B_t = 0$$

A.4.2 Model Solution

Rewriting the solution of the model given in (6), we have:

$$\begin{bmatrix} \pi_t \\ x_t \end{bmatrix} = -\sum_{j=0}^{\infty} A^{-(j+1)} \begin{bmatrix} -1/\beta & 0 & 0 \\ \sigma/\beta & \sigma & -\sigma \end{bmatrix} \begin{bmatrix} \mathbb{E}_t u_{t+j} \\ \mathbb{E}_t \epsilon_{t+j} \\ \mathbb{E}_t r_{t+j}^* \end{bmatrix}$$
(13)

With

$$A = \begin{bmatrix} \frac{1}{\beta} & -\frac{\kappa}{\beta} \\ \sigma \left(\phi_{\pi} - \frac{1}{\beta} \right) & 1 + \sigma \left(\phi_{x} + \frac{\kappa}{\beta} \right) \end{bmatrix}$$

Defining $B \equiv A^{-1}$ and $C \equiv A^{-2}$ and having only information about the current shocks and signals about one period ahead, we can write this solution as:

$$\begin{bmatrix} \pi_t \\ x_t \end{bmatrix} = -B \begin{bmatrix} -1/\beta & 0 & 0 \\ \sigma/\beta & \sigma & -\sigma \end{bmatrix} \begin{bmatrix} u_t \\ \epsilon_t \\ r_t^* \end{bmatrix} - C \begin{bmatrix} -1/\beta & 0 & 0 \\ \sigma/\beta & \sigma & -\sigma \end{bmatrix} \begin{bmatrix} \mathbb{E}_t u_{t+j} \\ \mathbb{E}_t \epsilon_{t+j} \\ \mathbb{E}_t r_{t+j}^* \end{bmatrix}$$
(14)

$$\omega_t = \Lambda^{\omega}_{\epsilon} \epsilon_t + \Lambda^{\omega}_u u_t + \Lambda^{\omega}_r r^*_t + \Lambda^{\omega}_{\epsilon'} \mathbb{E}_t(\epsilon_{t+1}) + \Lambda^{\omega}_{u'} \mathbb{E}_t(u_{t+1}) + \Lambda^{\omega}_{r^{*'}} \mathbb{E}_t(r^*_{t+1})$$

for $\omega = \{\pi, x, i\}.$

To calculate that, we have

$$B = A^{-1} = \frac{\beta}{1 + \sigma\phi_x + \kappa\sigma\phi_\pi} \begin{bmatrix} 1 + \sigma\left(\phi_x + \frac{\kappa}{\beta}\right) & \frac{\kappa}{\beta} \\ -\sigma\left(\phi_\pi - \frac{1}{\beta}\right) & \frac{1}{\beta} \end{bmatrix} = \frac{1}{1 + \sigma\phi_x + \kappa\sigma\phi_\pi} \begin{bmatrix} \beta + \sigma\left(\beta\phi_x + \kappa\right) & \kappa \\ -\sigma\left(\beta\phi_\pi - 1\right) & 1 \end{bmatrix}$$

$$C = B^{2} = \frac{1}{(1 + \sigma\phi_{x} + \kappa\sigma\phi_{\pi})^{2}} \begin{bmatrix} (\beta + \sigma(\beta\phi_{x} + \kappa))^{2} + \sigma\kappa(1 - \beta\phi_{\pi}) & \kappa(1 + \beta + \sigma(\beta\phi_{x} + \kappa)) \\ -\sigma(\beta\phi_{\pi} - 1)(1 + \beta + \sigma(\beta\phi_{x} + \kappa)) & 1 + \sigma\kappa - \beta\sigma\kappa\phi_{\pi} \end{bmatrix}$$

And defining the coefficient of the matrices as B_{ij} as the one of row *i* and column *j* of *B* (and the same with *C*), we have that the coefficients of inflation are:

$$\begin{split} \Lambda_{\epsilon}^{\pi} &= -\sigma B_{12} = -\frac{\sigma \kappa}{1 + \sigma \phi_x + \kappa \sigma \phi_\pi} \\ \Lambda_{u}^{\pi} &= \frac{1}{\beta} \left(B_{11} - \sigma B_{12} \right) = \frac{1 + \sigma \phi_x}{1 + \sigma \phi_x + \kappa \sigma \phi_\pi} \\ \Lambda_{r}^{\pi} &= \sigma B_{12} = \frac{\sigma \kappa}{1 + \sigma \phi_x + \kappa \sigma \phi_\pi} \\ \Lambda_{\epsilon'}^{\pi} &= -\sigma C_{12} = \frac{-\sigma \kappa (1 + \beta + \sigma (\beta \phi_x + \kappa))}{(1 + \sigma \phi_x + \kappa \sigma \phi_\pi)^2} \\ \Lambda_{u'}^{\pi} &= \frac{1}{\beta} \left(C_{11} - \sigma C_{12} \right) = \frac{(\beta + \sigma (\beta \phi_x + \kappa))(1 + \sigma \phi_x) - \sigma \kappa \phi_\pi}{(1 + \sigma \phi_x + \kappa \sigma \phi_\pi)^2} \\ \Lambda_{r^{\star'}}^{\pi} &= \sigma C_{12} = \frac{\sigma \kappa (1 + \beta + \sigma (\beta \phi_x + \kappa))}{(1 + \sigma \phi_x + \kappa \sigma \phi_\pi)^2} \end{split}$$

Coefficients for output:

$$\begin{split} \Lambda_{\epsilon}^{x} &= -\sigma B_{22} = -\frac{\sigma}{1 + \sigma \phi_{x} + \kappa \sigma \phi_{\pi}} \\ \Lambda_{u}^{x} &= \frac{1}{\beta} \left(B_{21} - \sigma B_{22} \right) = -\frac{\sigma \phi_{\pi}}{1 + \sigma \phi_{x} + \kappa \sigma \phi_{\pi}} \\ \Lambda_{r}^{x} &= \sigma B_{22} = \frac{\sigma}{1 + \sigma \phi_{x} + \kappa \sigma \phi_{\pi}} \\ \Lambda_{\epsilon'}^{x} &= -\sigma C_{22} = \frac{-\sigma (1 + \sigma \kappa - \beta \sigma \kappa \phi_{\pi})}{(1 + \sigma \phi_{x} + \kappa \sigma \phi_{\pi})^{2}} \\ \Lambda_{u'}^{x} &= \frac{1}{\beta} \left(C_{21} - \sigma C_{22} \right) = -\frac{\sigma \left[(\beta \phi_{\pi} - 1)(1 + \beta + \sigma \beta \phi_{x}) + 1 \right]}{\beta (1 + \sigma \phi_{x} + \kappa \sigma \phi_{\pi})^{2}} \\ \Lambda_{r^{*'}}^{x} &= \sigma C_{22} = \frac{\sigma (1 + \sigma \kappa - \beta \sigma \kappa \phi_{\pi})}{(1 + \sigma \phi_{x} + \kappa \sigma \phi_{\pi})^{2}} \end{split}$$

And for the interest rate:

$$\begin{split} \Lambda^{i}_{\epsilon} &= 1 + \phi_{\pi} \Lambda^{\pi}_{\epsilon} + \phi_{x} \Lambda^{x}_{\epsilon} = \frac{1}{1 + \sigma \phi_{x} + \kappa \sigma \phi_{\pi}} \\ \Lambda^{i}_{u} &= \phi_{\pi} \Lambda^{\pi}_{u} + \phi_{x} \Lambda^{x}_{u} = \frac{\phi_{\pi}}{1 + \sigma \phi_{x} + \kappa \sigma \phi_{\pi}} \\ \Lambda^{i}_{r} &= \phi_{\pi} \Lambda^{\pi}_{r} + \phi_{x} \Lambda^{x}_{r} = \frac{\sigma \phi_{x} + \sigma \kappa \phi_{\pi}}{1 + \sigma \phi_{x} + \kappa \sigma \phi_{\pi}} \\ \Lambda^{i}_{\epsilon'} &= \phi_{\pi} \Lambda^{\pi}_{s^{\epsilon}} + \phi_{x} \Lambda^{x}_{s^{\epsilon}} = -\frac{(1 + \sigma \kappa)(\sigma \kappa \phi_{\pi} + \sigma \phi_{x}) + \beta \sigma \kappa \phi_{\pi}}{(1 + \sigma \phi_{x} + \kappa \sigma \phi_{\pi})^{2}} \\ \Lambda^{i}_{u'} &= \phi_{\pi} \Lambda^{\pi}_{s^{u}} + \phi_{x} \Lambda^{x}_{s^{u}} = \frac{1}{(1 + \sigma \kappa)(\sigma \kappa \phi_{\pi} + \sigma \phi_{x}) + \beta \sigma \kappa \phi_{\pi}}{(1 + \sigma \phi_{x} + \kappa \sigma \phi_{\pi})^{2}} \\ \Lambda^{i}_{r^{*'}} &= \phi_{\pi} \Lambda^{\pi}_{s^{r}} + \phi_{x} \Lambda^{x}_{s^{r}} = \frac{(1 + \sigma \kappa)(\sigma \kappa \phi_{\pi} + \sigma \phi_{x}) + \beta \sigma \kappa \phi_{\pi}}{(1 + \sigma \phi_{x} + \kappa \sigma \phi_{\pi})^{2}} \end{split}$$

Given the distribution of the shocks: $\xi_t \sim N(0, \sigma_{\xi}^2)$, and $s_t^{\xi} = \xi_{t+1} + \zeta_t^{\xi}$, with $\zeta_t^{\xi} \sim N(0, \sigma_{s^{\xi}}^2)$ for $\xi = \{\epsilon, u, r^*\}$, we can write the general solution then as a function of the observable variables:

$$\omega_t = \Lambda^{\omega}_{\epsilon} \epsilon_t + \Lambda^{\omega}_u u_t + \Lambda^{\omega}_r r^*_t + \Lambda^{\omega}_{s^{\epsilon}} s^{\epsilon}_t + \Lambda^{\omega}_{s^{u}} s^{u}_t + \Lambda^{\omega}_{s^{r}} s^{r}_t$$

for $\omega = \{\pi, x, i\}.$

Define $\mathbb{E}(x)$ as the expectations before seeing the interest rate and $\mathbb{E}(x|i_t)$ as the one after seeing the interest rate. Then the expectation of the exogenous variables before and after are seeing the interest rate are:

$$\mathbb{E}(\omega_t) = \Lambda^{\omega}_{\epsilon} \mathbb{E}(\epsilon_t) + \Lambda^{\omega}_{u} \mathbb{E}(u_t) + \Lambda^{\omega}_{r} \mathbb{E}(r^*_t) + \Lambda^{\omega}_{s^{\epsilon}} \mathbb{E}(s^{\epsilon}_t) + \Lambda^{\omega}_{s^{u}} \mathbb{E}(s^{u}_t) + \Lambda^{\omega}_{s^{r}} \mathbb{E}(s^{r}_t)$$
(15)

$$\mathbb{E}(\omega_t|i_t) = \Lambda^{\omega}_{\epsilon} \mathbb{E}(\epsilon_t|i_t) + \Lambda^{\omega}_u \mathbb{E}(u_t|i_t) + \Lambda^{\omega}_r \mathbb{E}(r^*_t|i_t) + \Lambda^{\omega}_{s^{\epsilon}} \mathbb{E}(s^{\epsilon}_t|i_t) + \Lambda^{\omega}_{s^{u}} \mathbb{E}(s^{u}_t|i_t) + \Lambda^{\omega}_{s^{r}} \mathbb{E}(s^{\ell}_t|i_t) + \Lambda^{\omega}_{s^{r}}$$

A.4.3 Extracting information from the signal

The expectation of future shocks comes from a signal extraction problem. Defining $\gamma_{s\xi} = \sigma_{s\xi}^{-2}$ and $\gamma_{\xi} = \sigma_{\xi}^{-2}$ for $\xi = \{\epsilon, u, r^*\}$, we calculate the posterior distribution and expectations of the shock given the signal and the knowledge of the distributions:

$$p\left(\xi_{t+1}|s_{t}^{\xi}\right) \propto p\left(s_{t}^{\xi}/\xi_{t+1}\right)p\left(\xi_{t+1}\right)$$
$$\propto \exp\left\{-\frac{1}{2}\left(\frac{\left(s_{t}^{\xi}-\xi_{t+1}\right)^{2}}{\sigma_{s\xi}^{2}}+\frac{\xi_{t+1}^{2}}{\sigma_{\xi}^{2}}\right)\right\}$$
$$\propto \exp\left\{-\frac{1}{2}\left(\xi_{t+1}^{2}(\gamma_{s\xi}+\gamma_{\xi})-2s_{t}^{\xi}\xi_{t+1}\gamma_{s\xi}\right)\right\}$$
$$\propto \exp\left\{-\frac{1}{2\frac{1}{\gamma_{s\xi}+\gamma_{\xi}}}\left(\xi_{t+1}-\frac{s_{t}^{\xi}\gamma_{s\xi}}{\gamma_{s\xi}+\gamma_{\xi}}\right)^{2}\right\}$$

So, the expectation is $\mathbb{E}_t(\xi_{t+1}|s_t^{\xi}) = \frac{\gamma_{s\xi}}{\gamma_{s\xi}+\gamma_{\xi}}s_t^{\xi}$, implying that $\Lambda_{s\xi}^{\omega} = \frac{\gamma_{s\xi}}{\gamma_{s\xi}+\gamma_{\xi}}\Lambda_{\xi'}^{\omega}$ for $\xi = \{\epsilon, u, r^*\}$. Note that the simple of Λ^{ω} is the same of the simple of Λ^{ω} .

Note that the sign of $\Lambda_{s^{\xi}}^{\omega}$ is the same as the sign of $\Lambda_{\xi'}^{\omega}$.

A.4.4 Extracting Information About One Shock

When agents do not know the information of only one shock, the solution has two characteristics: (i) they get perfect information after seeing the interest rate; (ii) the only terms that survive the difference between the expectation after the interest rate and before the interest, (16)-(15), is the one of the shock in question. For example, take the case of the current monetary shock. Then we have:

$$\mathbb{E}(\omega_t|i_t) - \mathbb{E}(\omega_t) = \Lambda^{\omega}_{\epsilon}(\mathbb{E}(\epsilon_t|i_t) - \mathbb{E}(\epsilon_t))$$

since for the rest of the shocks $\mathbb{E}(\xi_t|i_t) = \mathbb{E}(\xi_t)$ for $\xi_t = \{u_t, r_t^*, \epsilon_{t+1}, u_{t+1}, r_{t+1}^*\}$.

Since we are looking for the coefficient that relates the difference of inflation and output expectations to the difference in expectations of the interest rate, we will have in this case:

$$\mathbb{E}(x_t|i_t) - \mathbb{E}(x_t) = \Lambda^x_{\epsilon}(\mathbb{E}(\epsilon_t|i_t) - \mathbb{E}(\epsilon_t))$$
$$\mathbb{E}(\pi_t|i_t) - \mathbb{E}(\pi_t) = \Lambda^\pi_{\epsilon}(\mathbb{E}(\epsilon_t|i_t) - \mathbb{E}(\epsilon_t))$$
$$\mathbb{E}(i_t|i_t) - \mathbb{E}(i_t) = i_t - \mathbb{E}(i_t) = \Lambda^i_{\epsilon}(\mathbb{E}(\epsilon_t|i_t) - \mathbb{E}(\epsilon_t))$$

and so the coefficients of interest are:

$$\mathbb{E}(x_t|i_t) - \mathbb{E}(x_t) = \chi_x(i_t - \mathbb{E}(i_t)) = \frac{\Lambda_{\epsilon}^x}{\Lambda_{\epsilon}^i}(i_t - \mathbb{E}(i_t))$$
$$\mathbb{E}(\pi_t|i_t) - \mathbb{E}(\pi_t) = \chi_\pi(i_t - \mathbb{E}(i_t)) = \frac{\Lambda_{\epsilon}^\pi}{\Lambda_{\epsilon}^i}(i_t - \mathbb{E}(i_t))$$

The same way, we analyze one shock each time and get the coefficients χ_{π} and χ_x of Table 2 in each case:

• With only the monetary Shock, ϵ_t :

$$\chi_{\pi} = -\sigma\kappa < 0 \qquad \qquad \chi_x = -\sigma < 0$$

• With only the cost push shock, u_t :

$$\chi_{\pi} = \frac{1 + \sigma \phi_x}{\phi_{\pi}} > 0 \qquad \qquad \chi_x = -\sigma < 0$$

• With only the Natural rate of interest, r_t^* :

$$\chi_{\pi} = \frac{\kappa}{\phi_{\pi}\kappa + \phi_x} > 0 \qquad \qquad \chi_x = \frac{1}{\phi_{\pi}\kappa + \phi_x} > 0$$
• With only a signal of future monetary shock, $s_t^\epsilon :$

$$\chi_{\pi} = \frac{\kappa(1+\beta+\sigma(\beta\phi_{x}+\kappa))}{\kappa\phi_{\pi}(1+\beta+\sigma\kappa)+\phi_{x}(1+\sigma\kappa)} > 0$$

$$\chi_{x} = \frac{1+\sigma\kappa(1-\beta\phi_{\pi})}{\kappa\phi_{\pi}(1+\beta+\sigma\kappa)+\phi_{x}(1+\sigma\kappa)} > 0 \quad \text{if } \phi_{\pi} < \frac{1+\sigma\kappa}{\beta\sigma\kappa}$$

• With only a signal of future cost push shock, s_t^u :

$$\chi_{\pi} = \frac{(\beta + \sigma(\beta\phi_x + \kappa))(1 + \sigma\phi_x) - \sigma\kappa\phi_{\pi}}{(\sigma\phi_x + \sigma\kappa\phi_{\pi})(1 + \sigma\phi_x - \phi_{\pi}) + \beta\phi_{\pi}(1 + \sigma\phi_x)}$$
$$\chi_x = -\sigma \frac{(\beta\phi_{\pi} - 1)(1 + \sigma\phi_x) + \phi_{\pi}}{(\sigma\phi_x + \sigma\kappa\phi_{\pi})(1 + \sigma\phi_x - \phi_{\pi}) + \beta\phi_{\pi}(1 + \sigma\phi_x)}$$

The condition for the denominators to be positive is:

$$\phi_{\pi} \text{ btw. } \frac{\sigma\phi_x - (1 + \sigma\phi_x)(\sigma\kappa + \beta) \pm \sqrt{[\sigma\phi_x - (1 + \sigma\phi_x)(\sigma\kappa + \beta)]^2 + 4\sigma^2\kappa\phi_x(1 + \sigma\phi_x)}}{-2\sigma\kappa}$$

which is always met under the coefficients described in the table below.

The condition for the numerator of the reaction of inflation to be positive is:

$$\phi_{\pi} < \frac{(\beta + \sigma(\beta\phi_x + \kappa))(1 + \sigma\phi_x)}{\sigma\kappa}$$

which is always met under the coefficients described in the table below.

The condition for the numerator of the reaction of output to be positive is:

$$\phi_{\pi} < \frac{1 + \sigma \phi_x}{1 + \beta (1 + \sigma \phi_x)}$$

which is never met and goes in conflict with the Taylor principle.

• With only a signal of future natural rate of interest, $s_t^{r^*}$:

$$\chi_{\pi} = \frac{\kappa(1+\beta+\sigma(\beta\phi_{x}+\kappa))}{\kappa\phi_{\pi}(1+\beta+\sigma\kappa)+\phi_{x}(1+\sigma\kappa)} > 0$$

$$\chi_{x} = \frac{1+\sigma\kappa(1-\beta\phi_{\pi})}{\kappa\phi_{\pi}(1+\beta+\sigma\kappa)+\phi_{x}(1+\sigma\kappa)} > 0 \quad \text{if } \phi_{\pi} < \frac{1+\sigma\kappa}{\beta\sigma\kappa}$$

The coefficients need to satisfy then the Taylor principle (written in the main part of

the text) and the conditions boxed above to be positive. For the cases that the conditions depended on specific values, the ranges that were analyzed were:

 TABLE 22: Parameter Values for Numerical Exercises

α	β	σ	κ	ϕ_x	ϕ_{π}
0.6 - 0.9	0.99	0.5 - 10	0.01 - 0.05	0 - 0.5	1 - 2.9

The sign for those coefficients were then the ones given in Table 2.

A.4.5 Extracting Information About Two Shocks

In this case, the information extracted comes from a signal extraction problem in which the agents get an expectation of two shocks each time after observing only one signal: the interest rate. For simplicity, it is assumed that there is no information besides the interest rate and there is full knowledge of the distributions and parameters. Each cases analyzed uses the following solutions of a signal extraction problem from normal distributions: Suppose the unobservable is $X_t \sim N(B_t, \Sigma)$ and the observable $S_t | X_t \sim N(AX_t, \Omega)$. The posterior is $X_t | S_t \sim N(\mu_t, \Gamma)$, with:

$$\mu_t = \Gamma(A'\Omega^{-1}S_t + \Sigma^{-1}B_t)$$
$$\Gamma^{-1} = (A'\Omega^{-1}A + \Sigma^{-1})$$

• Current monetary and natural rate of interest shocks: In this case, we can write the interest rate as $i_t = \Lambda_{\epsilon}^i \epsilon_t + \Lambda_r^i r_t^*$ and applying the previous definitions we have:

$$X_{t} = \begin{bmatrix} \epsilon_{t} \\ r_{t}^{*} \end{bmatrix} \qquad B_{t} = \begin{bmatrix} 0 \\ 0 \end{bmatrix} \qquad \Sigma = \begin{bmatrix} \sigma_{\epsilon}^{2} & 0 \\ 0 & \sigma_{r}^{2} \end{bmatrix}$$
$$S_{t} = i_{t} \qquad A = \begin{bmatrix} \Lambda_{\epsilon}^{i} & \Lambda_{r}^{i} \end{bmatrix} \qquad \Omega = \sigma_{\xi}^{2} \to 0$$

So, replacing and defining $\gamma_z = \sigma_z^{-2}$, the posterior mean and variance are:

$$\mu_t = \begin{bmatrix} \mu^{\epsilon} \\ \mu^r \end{bmatrix} i_t = \frac{1}{\gamma_r (\Lambda^i_{\epsilon})^2 + \gamma_{\epsilon} (\Lambda^i_{r})^2} \begin{bmatrix} \gamma_r \Lambda^i_{\epsilon} \\ \gamma_{\epsilon} \Lambda^i_{r} \end{bmatrix} i_t \qquad \Gamma = \frac{1}{\gamma_r (\Lambda^i_{\epsilon})^2 + \gamma_{\epsilon} (\Lambda^i_{r})^2} \begin{bmatrix} (\Lambda^i_{r})^2 & -\Lambda^i_{r} \Lambda^i_{\epsilon} \\ -\Lambda^i_{r} \Lambda^i_{\epsilon} & (\Lambda^i_{\epsilon})^2 \end{bmatrix}$$

With that in hand, we can now write the difference of expectations of output and inflation as:

$$\begin{split} \mathbb{E}(\omega_t|i_t) - \mathbb{E}(\omega_t) &= \Lambda^{\omega}_{\epsilon}(\mathbb{E}(\epsilon_t|i_t) - \mathbb{E}(\epsilon_t)) + \Lambda^{\omega}_{r}(\mathbb{E}(r_t^*|i_t) - \mathbb{E}(r_t^*)) \\ &= \Lambda^{\omega}_{\epsilon}\mu^{\epsilon}i_t + \Lambda^{\omega}_{r}\mu^{r}i_t \\ &= (\Lambda^{\omega}_{\epsilon}\mu^{\epsilon} + \Lambda^{\omega}_{r}\mu^{r})(i_t - \mathbb{E}(i_t)) \end{split}$$

And with this in hand the coefficients of interest are:

$$\begin{split} \chi_{\pi} &= \Lambda_{\epsilon}^{\pi} \mu^{\epsilon} + \Lambda_{r}^{\pi} \mu^{r} \\ &= \Lambda_{\epsilon}^{\pi} \left(\frac{\gamma_{r} \Lambda_{\epsilon}^{i}}{\gamma_{r} (\Lambda_{\epsilon}^{i})^{2} + \gamma_{\epsilon} (\Lambda_{r}^{i})^{2}} \right) + \Lambda_{r}^{\pi} \left(\frac{\gamma_{\epsilon} \Lambda_{r}^{\pi}}{\gamma_{r} (\Lambda_{\epsilon}^{i})^{2} + \gamma_{\epsilon} (\Lambda_{r}^{i})^{2}} \right) \\ &= \frac{\frac{-\sigma \kappa \gamma_{r} + \sigma \kappa \gamma_{\epsilon} (\sigma \phi_{x} + \sigma \kappa \phi_{\pi})}{(1 + \sigma \phi_{x} + \sigma \kappa \phi_{\pi})^{2}}}{\frac{\gamma_{r} + \gamma_{\epsilon} (\sigma \phi_{x} + \sigma \kappa \phi_{\pi})^{2}}{(1 + \sigma \phi_{x} + \sigma \kappa \phi_{\pi})^{2}}} \\ &= \frac{-\sigma \kappa \gamma_{r} + \sigma \kappa \gamma_{\epsilon} (\sigma \phi_{x} + \sigma \kappa \phi_{\pi})}{\gamma_{r} + \gamma_{\epsilon} (\sigma \phi_{x} + \sigma \kappa \phi_{\pi})^{2}} \end{split}$$

$$\chi_x = \Lambda_{\epsilon}^x \mu^{\epsilon} + \Lambda_r^x \mu^r$$

$$= \Lambda_{\epsilon}^x \left(\frac{\gamma_r \Lambda_{\epsilon}^i}{\gamma_r (\Lambda_{\epsilon}^i)^2 + \gamma_{\epsilon} (\Lambda_r^i)^2} \right) + \Lambda_r^x \left(\frac{\gamma_{\epsilon} \Lambda_r^x}{\gamma_r (\Lambda_{\epsilon}^i)^2 + \gamma_{\epsilon} (\Lambda_r^i)^2} \right)$$

$$= \frac{-\sigma \gamma_r + \sigma \gamma_{\epsilon} (\sigma \phi_x + \sigma \kappa \phi_\pi)}{\gamma_r + \gamma_{\epsilon} (\sigma \phi_x + \sigma \kappa \phi_\pi)^2}$$

Both are positive if

$$\boxed{\sigma_r^2(\sigma\phi_x + \sigma\kappa\phi_\pi) > \sigma_\epsilon^2}$$

• Current monetary and news about the future monetary shock: In this case, we can write the interest rate as $i_t = \Lambda^i_\epsilon \epsilon_t + \Lambda^i_{s^\epsilon} s^\epsilon_t$ and applying the previous definitions we have:

$$X_{t} = \begin{bmatrix} \epsilon_{t} \\ s_{t}^{\epsilon} \end{bmatrix} \qquad B_{t} = \begin{bmatrix} 0 \\ 0 \end{bmatrix} \qquad \Sigma = \begin{bmatrix} \sigma_{\epsilon}^{2} & 0 \\ 0 & \sigma_{s}^{2} + \sigma_{\epsilon}^{2} \end{bmatrix}$$
$$S_{t} = i_{t} \qquad A = \begin{bmatrix} \Lambda_{\epsilon}^{i} & \Lambda_{s}^{i} \end{bmatrix} \qquad \Omega = \sigma_{\xi}^{2} \to 0$$

So, replacing and defining $\gamma_z = \sigma_z^{-2}$, the posterior mean and variance are:

$$\mu_{t} = \begin{bmatrix} \mu^{\epsilon} \\ \mu^{s^{\epsilon}} \end{bmatrix} i_{t} = \frac{1}{(\Lambda^{i}_{\epsilon})^{2} \gamma_{s^{\epsilon}} + (\Lambda^{i}_{s^{\epsilon}})^{2} (\gamma_{\epsilon} + \gamma_{s^{\epsilon}})} \begin{bmatrix} \gamma_{s^{\epsilon}} \Lambda^{i}_{\epsilon} \\ (\gamma_{\epsilon} + \gamma_{s^{\epsilon}}) \Lambda^{i}_{s^{\epsilon}} \end{bmatrix} i_{t}$$

$$\Gamma = \frac{\gamma_{s^{\epsilon}} + \gamma_{\epsilon}}{(\Lambda^{i}_{\epsilon})^{2} \gamma_{s^{\epsilon}} \gamma_{\epsilon} + (\Lambda^{i}_{s^{\epsilon}})^{2} \gamma_{\epsilon} (\gamma_{s^{\epsilon}} + \gamma_{\epsilon})} \begin{bmatrix} (\Lambda^{i}_{s^{\epsilon}})^{2} & -\Lambda^{i}_{s^{\epsilon}} \Lambda^{i}_{\epsilon} \\ -\Lambda^{i}_{s^{\epsilon}} \Lambda^{i}_{\epsilon} & (\Lambda^{i}_{\epsilon})^{2} \end{bmatrix}$$

With that in hand, we can now write the difference of expectations of output and inflation as:

$$\begin{split} \mathbb{E}(\omega_t|i_t) - \mathbb{E}(\omega_t) &= \Lambda^{\omega}_{\epsilon}(\mathbb{E}(\epsilon_t|i_t) - \mathbb{E}(\epsilon_t)) + \Lambda^{\omega}_{s^{\epsilon}}(\mathbb{E}(s^{\epsilon}_t|i_t) - \mathbb{E}(s^{\epsilon}_t)) \\ &= \Lambda^{\omega}_{\epsilon}\mu^{\epsilon}i_t + \Lambda^{\omega}_{s^{\epsilon}}\mu^{s^{\epsilon}}i_t \\ &= (\Lambda^{\omega}_{\epsilon}\mu^{\epsilon} + \Lambda^{\omega}_{s^{\epsilon}}\mu^{s^{\epsilon}})(i_t - \mathbb{E}(i_t)) \end{split}$$

And with this in hand the coefficients of interest are:

$$\chi_{\pi} = \frac{\Lambda_{\epsilon}^{\pi} \Lambda_{\epsilon}^{i} \gamma_{s^{\epsilon}} + \Lambda_{s^{\epsilon}}^{\pi} \Lambda_{s^{\epsilon}}^{i} (\gamma_{s^{\epsilon}} + \gamma_{\epsilon})}{(\Lambda_{\epsilon}^{i})^{2} \gamma_{s^{\epsilon}} + (\Lambda_{s^{\epsilon}}^{i})^{2} (\gamma_{\epsilon} + \gamma_{s^{\epsilon}})}$$

This is positive if:

$$\underbrace{\underbrace{\Lambda_{\epsilon}^{\pi}}_{<0} \underbrace{\Lambda_{\epsilon}^{i}}_{<0} \underbrace{(\gamma_{s^{\epsilon}}}_{>0} + \underbrace{\Lambda_{s^{\epsilon}}^{\pi}}_{<0} \underbrace{\Lambda_{s^{\epsilon}}^{i}}_{<0} \underbrace{(\gamma_{s^{\epsilon}} + \gamma_{\epsilon})}_{>0} > 0$$

And

$$\chi_x = \frac{\Lambda_{\epsilon}^x \Lambda_{\epsilon}^i \gamma_{s^{\epsilon}} + \Lambda_{s^{\epsilon}}^x \Lambda_{s^{\epsilon}}^i (\gamma_{s^{\epsilon}} + \gamma_{\epsilon})}{(\Lambda_{\epsilon}^i)^2 \gamma_{s^{\epsilon}} + (\Lambda_{s^{\epsilon}}^i)^2 (\gamma_{\epsilon} + \gamma_{s^{\epsilon}})}$$

This is positive if:

$$\underbrace{\underbrace{\Lambda_{\epsilon}^{x}}_{<0} \underbrace{\Lambda_{\epsilon}^{i}}_{<0} \underbrace{(\gamma_{s^{\epsilon}})}_{>0} + \underbrace{\Lambda_{s^{\epsilon}}^{x}}_{<0 \text{ if } \phi_{\pi} < \frac{1 + \sigma \kappa}{\beta \sigma \kappa}} \underbrace{\Lambda_{s^{\epsilon}}^{i}}_{<0} \underbrace{(\gamma_{s^{\epsilon}} + \gamma_{\epsilon})}_{>0} > 0}_{>0 \text{ if } \phi_{\pi} < \frac{1 + \sigma \kappa}{\beta \sigma \kappa}}$$

The reaction of inflation then is sometimes positive depending on the model's parameters (happens when σ , ϕ_x , κ , ϕ_π are relatively high). The output reaction is negative for all the parameter values described in Table 22 and any combination of the standard deviation of the shocks. The reason is that even if the equations generally meet the condition $\phi_\pi < \frac{1+\sigma\kappa}{\beta\sigma\kappa}$, it is impossible to increase the importance of the second factor, which is the positive one, relative to the first. The reason behind this is that they are both from the same shock (cannot decrease γ_{ϵ} below zero and changes in $\gamma_{s^{\epsilon}}$ affect both terms at the same time.

• Current monetary and news about the future natural rate shock: In this case, we can write the interest rate as $i_t = \Lambda_{\epsilon}^i \epsilon_t + \Lambda_{s^r}^i s_t^r$ and applying the previous definitions we have:

$$X_{t} = \begin{bmatrix} \epsilon_{t} \\ s_{t}^{r} \end{bmatrix} \qquad B_{t} = \begin{bmatrix} 0 \\ 0 \end{bmatrix} \qquad \Sigma = \begin{bmatrix} \sigma_{\epsilon}^{2} & 0 \\ 0 & \sigma_{s^{r}}^{2} + \sigma_{r}^{2} \end{bmatrix}$$
$$S_{t} = i_{t} \qquad A = \begin{bmatrix} \Lambda_{\epsilon}^{i} & \Lambda_{s^{r}}^{i} \end{bmatrix} \qquad \Omega = \sigma_{\xi}^{2} \to 0$$

So, replacing and defining $\gamma_z = \sigma_z^{-2}$, the posterior mean and variance are:

$$\mu_{t} = \begin{bmatrix} \mu^{\epsilon} \\ \mu^{s^{r}} \end{bmatrix} i_{t} = \frac{1}{(\Lambda^{i}_{\epsilon})^{2} \gamma_{r} \gamma_{s^{r}} + (\Lambda^{i}_{s^{r}})^{2} (\gamma_{r} + \gamma_{s^{r}}) \gamma_{\epsilon}} \begin{bmatrix} \gamma_{r} \gamma_{s^{r}} \Lambda^{i}_{\epsilon} \\ \gamma_{\epsilon} (\gamma_{r} + \gamma_{s^{r}}) \Lambda^{i}_{s^{r}} \end{bmatrix} i_{t}$$

$$\Gamma = \frac{\gamma_{s^{r}} + \gamma_{r}}{(\Lambda^{i}_{\epsilon})^{2} \gamma_{s^{r}} \gamma_{r} + (\Lambda^{i}_{s^{r}})^{2} \gamma_{\epsilon} (\gamma_{s^{r}} + \gamma_{r})} \begin{bmatrix} (\Lambda^{i}_{s^{r}})^{2} & -\Lambda^{i}_{s^{r}} \Lambda^{i}_{\epsilon} \\ -\Lambda^{i}_{s^{r}} \Lambda^{i}_{\epsilon} & (\Lambda^{i}_{\epsilon})^{2} \end{bmatrix}$$

With that in hand, we can now write the difference of expectations of output and

inflation as:

$$\begin{split} \mathbb{E}(\omega_t|i_t) - \mathbb{E}(\omega_t) &= \Lambda^{\omega}_{\epsilon}(\mathbb{E}(\epsilon_t|i_t) - \mathbb{E}(\epsilon_t)) + \Lambda^{\omega}_{s^r}(\mathbb{E}(s^r_t|i_t) - \mathbb{E}(s^r_t)) \\ &= \Lambda^{\omega}_{\epsilon}\mu^{\epsilon}i_t + \Lambda^{\omega}_{s^r}\mu^{s^r}i_t \\ &= (\Lambda^{\omega}_{\epsilon}\mu^{\epsilon} + \Lambda^{\omega}_{s^r}\mu^{s^r})(i_t - \mathbb{E}(i_t)) \end{split}$$

And with this in hand the coefficients of interest are:

$$\chi_{\pi} = \frac{\Lambda_{\epsilon}^{\pi} \Lambda_{\epsilon}^{i} \gamma_{r} \gamma_{s^{r}} + \Lambda_{s^{r}}^{\pi} \Lambda_{s^{r}}^{i} \gamma_{\epsilon} (\gamma_{s^{r}} + \gamma_{r})}{(\Lambda_{\epsilon}^{i})^{2} \gamma_{r} \gamma_{s^{r}} + (\Lambda_{s^{r}}^{i})^{2} (\gamma_{r} + \gamma_{s^{r}}) \gamma_{\epsilon}}$$

This is positive if:

$$\underbrace{\underbrace{\Lambda_{\epsilon}^{\pi}}_{<0} \underbrace{\Lambda_{\epsilon}^{i}}_{<0} \underbrace{(\gamma_{s^{r}} \gamma_{r})}_{>0} + \underbrace{\Lambda_{s^{r}}^{\pi}}_{>0} \underbrace{\Lambda_{s^{r}}^{i}}_{>0} \underbrace{\gamma_{\epsilon}(\gamma_{s^{r}} + \gamma_{r})}_{>0} > 0}_{>0}$$

$$\chi_x = \frac{\Lambda_{\epsilon}^x \Lambda_{\epsilon}^i \gamma_r \gamma_{s^r} + \Lambda_{s^r}^x \Lambda_{s^r}^i \gamma_{\epsilon} (\gamma_{s^r} + \gamma_r)}{(\Lambda_{\epsilon}^i)^2 \gamma_r \gamma_{s^r} + (\Lambda_{s^r}^i)^2 (\gamma_r + \gamma_{s^r}) \gamma_{\epsilon}}$$

This is positive if:

$$\underbrace{\underbrace{\Lambda_{\epsilon}^{x}}_{<0} \underbrace{\Lambda_{\epsilon}^{i}}_{<0} \underbrace{(\gamma_{s^{r}} \gamma_{r})}_{>0} + \underbrace{\Lambda_{s^{r}}^{x}}_{\geq 0 \text{ if } \phi_{\pi} < \frac{1 + \sigma \kappa}{\beta \sigma \kappa}} \underbrace{\Lambda_{s^{r}}^{i}}_{>0} \underbrace{\gamma_{\epsilon}(\gamma_{s^{r}} + \gamma_{r})}_{>0} > 0}_{>0}$$

Similarly to the previous case, the output's reaction can become positive first if the parameters satisfy $\phi_{\pi} < \frac{1+\sigma\kappa}{\beta\sigma\kappa}$. And then, for both to become positive, we need to increase the importance of the part about the news about the natural rate. The positive part will more than outweigh the negative one when the variability of the natural rate is relatively high (precision low), and the variability of the monetary shock is relatively low (precision high). The conclusion is then similar to the case

of current monetary and natural rate shocks, but the condition to satisfy a positive value is even more challenging for output.

Documentos de Trabajo Banco Central de Chile	Working Papers Central Bank of Chile		
NÚMEROS ANTERIORES	PAST ISSUES		
La serie de Documentos de Trabajo en versión PDF puede obtenerse gratis en la dirección electrónica:	Working Papers in PDF format can be downloaded free of charge from:		
www.bcentral.cl/esp/estpub/estudios/dtbc.	www.bcentral.cl/eng/stdpub/studies/workingpaper.		
Existe la posibilidad de solicitar una copia impresa con un costo de Ch\$500 si es dentro de Chile y US\$12 si es fuera de Chile. Las solicitudes se pueden hacer por fax: +56 2 26702231 o a través del correo electrónico: <u>bcch@bcentral.cl</u> .	Printed versions can be ordered individually for US\$12 per copy (for order inside Chile the charge is Ch\$500.) Orders can be placed by fax: +56 2 26702231 or by email: <u>bcch@bcentral.cl</u> .		

DTBC – 1017 Is the Information Channel of Monetary Policy Alive in Emerging Markets? Mariana García-Schmidt

DTBC – 1016 **The Portfolio Choice Channel of Wealth Inequality** Mauricio Calani, Lucas Rosso

DTBC – 1015 Fiscal Consolidations in Commodity-Exporting Countries: A DSGE Perspective Manuel González-Astudillo, Juan Guerra-Salas, Avi Lipton

DTBC – 1014 Accounting for Nature in Economic Models Nicoletta Batini, Luigi Durand

DTBC – 1013 **Transmission Mechanisms in HANK: an Application to Chile** Benjamín García, Mario Giarda, Carlos Lizama, Ignacio Rojas

DTBC – 1012 Cyclical wage premia in the informal labour market: Persistent and downwardly rigid Daniel Guzmán DTBC – 1011 Macro Implications of Inequality-driven Political Polarization Alvaro Aguirre

DTBC – 1010 Firm Shocks, Workers Earnings and the Extensive Margin Álvaro Castillo, Ana Sofía León, Antonio Martner, Matías Tapia

DTBC – 1009 Trade Policy and Reallocation: Multinational vs. Single-Country Linkages in the Tire Industry Brian Pustilnik

DTBC - 1008

The finances of Chilean households during the pandemic: an assessment from the 2021 Household Financial Survey

Enzo Cerletti, Magdalena Cortina, Alejandra Inzunza, Felipe Martínez, Patricio Toro

DTBC - 1007

UIP Deviations: Insights from Event Studies Elias Albagli, Luis Ceballos, Sebastian Claro, Damian Romero

DTBC – 1006 Input price dispersion across buyers and misallocation Ariel Burstein, Javier Cravino, Marco Rojas

DTBC – 1005 The Role of Parental Altruism in Parents Consumption, College Financial Support, and Outcomes in Higher Education Agustín Díaz Casanueva

DTBC – 1004 Indebtedness and labor risk sorting across consumer lender types: evidence from Chile Carlos Madeira



DOCUMENTOS DE TRABAJO Junio 2024